# **AGREEMENT**

### **BETWEEN**

### CHEMTURA CANADA CO/CIE Elmira, Ontario

### **AND**

UNITED STEELWORKERS LOCAL UNION NO. 13691-00 Elmira, Ontario.

**June 2015 – May 2018** 

02402 (10)

### **PREAMBLE**

UNITED STEELWORKERS, LOCAL UNION 13691-00, hereinafter referred to as the Union, being certified by the Ontario Labour Relations Board to represent certain employees of CHEMTURA CANADA CO/CIE, hereinafter referred to as the Company, recognize that there exists between both parties a community of interests which consists of the support of all measures which tend to maintain harmonious and co-operative relations between both parties, which will in turn enhance production, working conditions, hours of work, wages and the general well-being of all concerned.

The Union encourages and will co-operate with the Company in any attempt to improve efficiency, lower production costs, eliminate waste, conserve material and supplies, and strengthen goodwill between the Company and its employees, the customer and the public.

We herewith freely enter into this Agreement with the intention of fostering a spirit of friendliness and mutual consideration which shall prevail in and around the properties of the Company in Elmira, Ontario.

NOTE: For the purpose of this Agreement, the masculine gender shall be applicable to both female and male employees, unless otherwise specified.

### NON-DISCRIMINATION AND HARASSMENT POLICY

### **PURPOSE**

Chemtura Canada Co./Cie and UNITED STEELWORKERS, LO-CAL UNION 13691-00 in recognition of the dignity and worth of every person will strive to provide equal rights and opportunities for everyone within an environment free of sexual harassment or discriminatory practices.

### NON-DISCRIMINATION

It is this Company's and Union's philosophy and policy to treat employees and applicants for employment fairly and without regard to race, colour, sex, religion, national origin, age, sexual orientation or any other unlawful consideration in accordance with the provisions of the Ontario Human Rights Code, as amended from time to time.

This policy applies to all employment practices including recruiting, hiring, pay rates, training and development, promotions, terminations and other terms and conditions of employment.

### HARASSMENT/RACIAL HARASSMENT/ WORKPLACE VIOLENCE

Chemtura Canada Co./Cie and UNITED STEELWORKERS, LO-CAL UNION 13691-00 will not tolerate any form of harassment, joking remarks or any course of conduct or comment which is known or ought reasonably to have been known to be unwelcome because of race, colour, creed, ancestry, place of origin or any other unlawful consideration in accordance with the provisions of the Ontario Human Rights Code, as amended from time to time. Workplace Violence and Workplace Harassment will not be tolerated as prohibited by the provisions of the Occupational Health and Safety Act.

### SEXUAL HARASSMENT

Chemtura Canada Co./Cie and UNITED STEELWORKERS, LO-CAL UNION 13691-00 prohibits sexual harassment of any employee. Sexual harassment is prohibited because it is intimidating and an abuse of power. It is defined as unwelcome sexual attention or advances, requests for sexual favours, and other verbal or physical conduct of a sexual nature.

### EMPLOYEE RECOURSE

Any employee who feels subject to discrimination or harassment should immediately report it to Management and/or Union Executive. Any reports or complaints made to the Company will be referred to the Union Executive by the Company. No retaliation against the employee will take place. Such reports will be investigated thoroughly by Company and Union Representatives. If the report has merit, disciplinary action will be taken against the offender. Depending on the severity of the misconduct, the disciplinary action could range from a warning to termination.

# ALPHABETICAL INDEX COLLECTIVE LABOUR AGREEMENT

Subject	Article Page
Arbitration	37
Bereavement Pay	47
Designated Holidays	715
Discipline	1438
Grievance Procedure	35
Hours of Work	4 6
Job Rates and Promotions	41
Jury and Witness Service	48
Overtime	5 7
Privacy of Employee Information	1540
Recognition	1 1
Renewal and Termination	1948
Representation	2 2
Safety and Health	25
Seniority	27
Shift Premiums	614
Transfers & Job Postings	818
Union Security	3 4
Vacations	921
Appendix "A" – Interim Increase	51
LOUs	
Payment for Negotiations	52
Safety Footwear	53
Blow the Whistle	54
Payroll Error Correction	55
Training	56
Committees, Teams or Events	57
"OSHA Supervisor" Indemnification	58
Coverage for Chief Operating Engineer	59
Employee Assistance Program	
Return to Work Program	64
Review of WERA AND Group RRSP Language	re 66

### AGREEMENT

### between

### CHEMTURA CANADA CO/CIE Elmira, Ontario

### and

### UNITED STEELWORKERS LOCAL UNION NO. 13691-00 Elmira, Ontario

This Agreement made and entered into this May 29, 2015 by and between CHEMTURA CANADA CO./CIE, of Elmira, Ontario, hereinafter referred to as the Company, and LOCAL UNION NO. 13691-00, UNITED STEELWORKERS, of Elmira, Ontario, hereinafter referred to as the Union.

The parties agree as follows:

### ARTICLE 1 RECOGNITION

- 1.01 The Union recognizes the Company's right to manage the plant and direct the working force which includes the right to hire, transfer, promote, demote, suspend, discipline or discharge for proper cause, to relieve employees of their duties because of lack of work or any other legitimate reason provided that no action shall be taken under this article which is in violation of the terms of this Agreement or is used in discrimination against any employee.
- 1.02 The Company recognizes UNITED STEELWORKERS, LOCAL UNION NO. 13691-00, as the sole collective bargaining agent for all hourly paid employees of the Company save and except Security Guards, and Department and Shift Supervisors, and those above the rank of supervisors, and office staff.
  - Students employed during their vacation or co-operative educational plan work periods are not covered by this Agreement unless employed on a job in the bargaining unit.
- 1.03 Both parties agree to abide by the laws with reference to Labour Relations presently in effect which define that – no

employer who is party to a collective agreement shall declare or cause a lock-out and no employee bound thereby shall go on strike unlawfully, cause a slow-down, sit-down or work stoppage during the term of the collective agreement. In the event of an unauthorised slow-down, sit-down, or work stoppage, partial or complete, the Union undertakes to co-operate actively with the Company to restore normal production.

1.04 A Department Head, Supervisor or Shift Supervisor, shall not perform work which supplants any other production or service employee on his regular job, except under circumstances requiring emergency or experimental work. Experimental work is defined as processes or equipment that remain in the scope of the Research and Development Department or Project and/or Process Engineers.

This shall not, however, prohibit qualified persons outside the bargaining unit from training an employee on an operation, or non-bargaining unit employees from taking inventory.

1.05 For purposes of this Agreement, the term "process" shall mean individual processes. The term "department" shall refer to the entire section, such as "Production Department", "Laboratory Technicians Department", "Trades Department".

### ARTICLE 2 REPRESENTATION

- **2.01** Union committees recognized by the Company are:
  - an Executive Committee of five (5) employees;
  - a Bargaining Unit Committee of three (3) employees;
  - a Grievance Committee of four (4) employees

The Bargaining Unit Committee is responsible for handling the day to day affairs of the bargaining unit employees of the Company. The Committee will consist of a Unit President, a Unit Secretary, and a Unit Grievor (Chief Steward).

At the discretion of either party, a staff representative of the United Steelworkers may accompany the Bargaining Unit Committee at a meeting of the parties, or the Grievance Committee at a third Step grievance meeting.

2.02 The Company shall recognize, from the areas where bargaining unit employees are working, up to a total of ten (10) Union Stewards, as assigned by the Union. The Company shall also recognize one (1) Benefits Officer and one (1) Compensation Officer. Such a Steward or Officer may, on receiving permission from his Department Head or Supervisor, or from the Shift Supervisor in their absence, leave his regular duties to investigate any grievance which may arise, without loss of wages.

A Compensation Officer will be recognized and assigned to matters related to WSIB and Return to Work. The Benefits Officer will be recognized and assigned to matters related to Employees Assistance Program, Employment Insurance, Group Life and Health Insurance Plan and any other specific problem deemed necessary by the Local Union.

- 2.03 The Union agrees to notify the Company promptly of any changes in Bargaining Unit representation.
- 2.04 The Company and the Bargaining Unit representatives shall meet when it is considered necessary by either party to discuss, negotiate, and amicably settle any difference, or other matter relating to this Agreement.

As defined in Article 2.01, payment will be made at straight time rate to an employee who is a member of the Bargaining Unit Committee for time spent at a meeting called by either party for discussion of matters related to administration of Agreements between the parties, and to a member of the Grievance Committee for time spent at a meeting held under the provisions of Article 12.01 for negotiation of a grievance up to but not including arbitration proceedings.

In addition to members of the Grievance Committee, not more than two employees involved in the dispute and a Department or Shift Steward for a total of not more than three additional employees may attend a meeting for negotiation of a grievance, and will be paid at straight time rate for time spent at the meeting.

2.05 On request, the Company will permit an employee to be absent from work without pay to perform duties for the local union, except that not more than five employees nor more than two employees from one job classification will be per-

mitted to be absent at any one time, unless agreed by the Company. The local union will submit lost time pay vouchers to the Company for employees on local union business. The Company will, in the employee's regular pay period, pay the affected employee the amount stipulated on the union lost time voucher subject to the usual deductions. The Company will invoice the local union for the amount of each lost time voucher and the local union will pay the Company within 30 calendar days.

It is understood that requested absence in excess of one week for more than two employees will be considered a request for leave of absence and subject to the provisions of Article 11.04.

2.06 The Company agrees to provide a bulletin board for the use of the Union to post Union notices. The Union agrees to inform the Human Resources Manager of all notices to be posted.

### ARTICLE 3 UNION SECURITY

- 3.01 Any person hired as a new employee or an employee transferred into the bargaining unit on or after the effective date of this Agreement will be required to sign the Authorization for Deduction Form. The Company agrees that all employees shall become and remain members of the Union as a condition of their continued employments.
- 3.02 The Company shall deduct Union dues including, where applicable, initiation fees and assessments, on a pay period basis, from the wages of each employee covered by this agreement. The amount of dues shall be calculated in accordance with the Union's Constitution. It is the responsibility of the Union to notify the Company of any change in the amount of deductions. Providing notice is received not later than the 20th day of the month, the change in the amount of deduction will become effective in the first pay period in the following month.
- 3.03 All dues, initiation fees and assessments shall be remitted to the Union forthwith and in any event no later than 15 days following the last day of the month in which the remittance was deducted. The remittance shall be sent to the Inter-

national Secretary Treasurer of the United Steelworkers, AFL-CIO-CLC, P.O. Box 13083 Postal Station "A", Toronto, Ontario M5W 1V7 in such form as shall be directed by the Union to the Company along with a completed Dues Remittance Form R-115. A copy of the Dues Remittance Form R-115 will also be sent to the Union office designated by the Area Coordinator.

- **3.04** The remittance and the R-115 form shall be accompanied by a statement containing the following information:
  - A list of the names of all employees from whom dues were deducted and the amount of dues deducted
  - A list of the names of all employees from whom no deductions have been made and reasons
  - c) This information shall be sent to both Union addresses identified in Article 3.03 in such form as shall directed by the Union to the Company
- **3.05** The Union shall indemnify and save the Company harmless against all claims, suits, judgments, attachments and from any other forms of liability that may arise out of any actions taken by the Company in compliance with this Article.
- **3.06** The Company, when preparing T-4 slips for the employees, will enter the amount of Union dues paid by the employee during the previous year.
- 3.07 The Company agrees to deduct on a pay period basis, the amount of one cent (\$.01) per hour from the wages of all employees in the bargaining unit for all hours worked and prior to the 15th day of the month following, to pay the amount so deducted to the "Humanity Fund" and to forward such payment to United Steelworkers National Office, 234 Eglinton Avenue East, Toronto, Ontario, M4P 1K7, and to advise in writing both the Humanity Fund at the aforementioned address and the Union that such payment has been made, the amount of such payment and the names of all employees in the bargaining unit on whose behalf the payment has been made. All employee deductions are voluntary and may be cancelled upon written request.
- 3.08 Employees shall ensure that the Company has current information for the Employee's address and phone numbers.

### ARTICLE 4 HOURS OF WORK

- 4.01 The normal daily and weekly hours of work are listed solely for the purpose of calculating overtime. It shall not be construed as indicating a guarantee of minimum hours, nor a restriction on the maximum daily or weekly hours to be worked.
- **4.02** The regular work week shall be as follows:
  - a) The workweek for an eight (8) hour shift schedule is defined as the seven-day period beginning and ending at 24:00 Saturday.
  - The workweek for a twelve (12) hour shift schedule is defined as a seven-day period beginning and ending at 05:30 Sunday.
- **4.03** Normal daily hours of work will be eight (8) hours per day and scheduled Monday to Friday.

These normal daily hours of work may be extended to include Saturday and/or Sunday, to meet production demands.

Normal daily hours of work will be twelve (12) hours per day scheduled Sunday to Saturday. The twelve (12) hour shift schedule will be day shift 05:30 to 17:30 and/or night shift 17:30 to 05:30.

**4.04** If considered necessary, the Company may establish two or three rotating eight (8) hour shifts on a 5-day schedule; or a twelve (12) hour rotating shift schedule and assign maintenance trades and laboratory technicians to work these shifts and/or schedules as required.

The Union will be notified in writing of any permanent change in an employee's normal daily hours of work.

4.05 A rotating shift worker will remain at his process until relieved by the employee taking his place on the next shift, or until other arrangements are made in case he does not appear.

Employees who must shower due to hygiene requirements will be paid \$5.75 per shift.

### 4.06 REST AND LUNCH PERIODS

Employees on the twelve (12) hour shift schedule will be permitted two (2) rest periods not to exceed fifteen (15)

minutes and one (1) lunch period not to exceed thirty (30) minutes on each shift at times scheduled by the Company.

All other employees will be permitted one (1) rest period not to exceed fifteen (15) minutes, and one (1) lunch period not to exceed twenty (20) minutes on each shift at times scheduled by the Company.

4.07 In the fall, when clocks revert back one (1) hour to Eastern Standard Time, the shifts affected by the change will both work 12.5-hour shifts. The night shift will work until 05:00 and the day shift will start at 05:00. The applicable premium rate will apply to the extra one-half hour worked by each affected shift.

In the spring when clocks are moved ahead to Daylight Savings Time, the shifts affected by the change will both work 11.5-hour shifts. The night shift will work until 06:00 and the day shift will start at 06:00.

- 4.08 An employee reporting for work on his regular shift or approved overtime shift who has not been informed that work was not available, shall receive four (4) hours pay at his applicable rate and be allowed to go home, unless the reason for work not being available is due to reasons over which the Company has no control. If, after reporting for work in such instances, it is possible for the Company to provide other work for the employee, he shall accept the work provided at his applicable rate of pay.
- 4.09 No payment shall be made to an employee reporting for work or remaining on the premises when no work is available as a result of a work stoppage by the Union members.
- 4.10 The payment of premium rates for Saturday and Sunday work is not intended to encourage employees to absent themselves from work on their regular scheduled shifts and the Union in no way condones such practice, and will co-operate with Management to eliminate such practice.

## ARTICLE 5 OVERTIME

- **5.01** (a) Overtime at one and one half times the individual employee's regular straight time rate will be paid for:
  - (1) Hours worked prior or subsequent to the normal daily hours of work outlined in Article 4.

(2) Hours worked on the first shift following a change of shift unless the employee has been notified 24 hours prior to the starting time of the new shift.

The provisions of (2) do not apply if the change of shift or change of schedule is the result of a work shortage.

(b) Employees on the 8 hour shift schedule will be paid as follows for all hours worked outside of his regularly scheduled shifts:

A rate of one and one half times the individual employee's regular straight time rate for all hours worked on a Saturday.

A rate of two times the individual employee's regular straight time rate for all hours worked on Sunday, or a Designated Holiday.

(c) Employees on the 12 hour shift schedule will be paid as follows for all hours worked outside of his regularly scheduled shifts:

A rate of one and a half times the individual employee's regular straight time rate will be paid for hours worked 05:30 Monday to 05:30 Sunday.

A rate of two times the individual employee's regular straight time rate for shifts commencing at 05:30 and 17:30 on Sunday, or a Designated Holiday.

- 5.02 (a) A CALL BACK to work is defined as unscheduled overtime on work assignments where an employee is required to return to work outside his normal working hours.
  - (1) Payment for such overtime shall be at the rate of one and one half times the employee's straight time rate of pay for all hours worked Monday through Saturday, and at double time the employee's straight time rate of pay for all hours worked on Sunday or on a Designated Holiday.
  - (2) A four (4) hour minimum allowance will be paid to all such employees having complied with a call back.
  - (3) Overtime assigned less than ten (10) minutes before the employee's normal quitting time will be

- considered to be a call back and provisions (1) and (2) of this Article 5.02 (a) will apply.
- (4) Assigned overtime of more than one hour prior to normal hours will be considered a call back and provisions (1) and (2) of this Article 5.02 (a) will apply. Such overtime shall be continuous with normal working hours and notification shall be given prior to 9:00 p.m. the previous evening.
- (5) Where an employee has complied with a call back assignment additional work assignments will not normally be made. In the event an additional work assignment is made, the employee will be considered on a second call back and the provisions (1) and (2) of this Article 5.02 (a) will apply. Except that if more than one assignment is made at the time of call back, the provisions of this sub article (5) will not apply.
- (6) A call back shall not apply to overtime continuance of unfinished work interrupted by an employee leaving the plant voluntarily and with permission.
- (b) SCHEDULED OVERTIME is defined as overtime scheduled prior to ten (10) minutes before the employee's normal quitting time and is not in conjunction with normal hours.
  - (1) Payment for such overtime shall be at the rate of one and one half times the employee's straight time rate of pay for all hours worked Monday through Saturday, and at double time the employee's straight time rate of pay for all hours worked on Sunday or on a Designated Holiday.
  - (2) A three (3) hour minimum allowance will be paid to all such employees having complied with the scheduled overtime.
  - (3) There shall be no limitation as to number or type of work assignments that may be made or assigned for such overtime.
- (c) CONTINUOUS OVERTIME is defined as overtime worked by an employee in conjunction with normal hours.

- Payment for hours worked on such overtime will be as provided in Article 5.01 (a) or (b) as applicable.
- (2) The provisions of sub article 5.01 (a) or (b), as applicable, shall apply to overtime continuance of unfinished work interrupted by an employee leaving the plant voluntarily and with permission.
- (3) When an employee is required to remain at his work station beyond his normal hours, he shall be paid as provided under sub article 5.01 (a) or (b), as applicable, until relieved by the employee taking his place on the next shift or until other arrangements are made in case the employee does not appear.
- 5.03 Employees who are required to work on Sunday, shall not be required to take time off in the same week to off set such Sunday work except to comply with the maximum weekly hours of work.
- **5.04** There shall be no duplication of overtime and other bonuses.
- **5.05** No overtime will be paid in the following cases:
  - (a) Change or exchange of hours at the request of employees with permission of the supervisor.
  - (b) Hours on which overtime or premium pay has been made shall not be used to calculate hours of work for further payment of overtime.

### 5.06 Plant Wide Overtime Procedures

(a) Overtime work to be performed in any job classification shall be performed by employees in the particular job classification concerned. Employees of the job classification are to be offered overtime first. Any such overtime work shall be shared equally, as far as practical, among the employees of the job classification. If no employees in that classification are available, overtime will then be offered to employees outside the classification who are qualified to perform the work consistent with the Company's policies and procedures. In the event that no employees are readily available, outside contractors may be called. This does not apply to work performed under Capital Projects.

### (b) Introduction

- All scheduled, continuous and call-back overtime is voluntary.
- 2. All scheduled, continuous and call-back overtime worked will be tracked and recorded.
- Double time overtime, time and one-half overtime, call back time and one-half overtime, and call-back double time overtime will be listed in separate categories.
- All double time overtime ranking will zero on the first Sunday of each calendar year with seniority being the deciding factor to start.
- All time and one-half overtime will zero every three (3) months on the first Sunday of the month with seniority being the deciding factor to start.
- All call-back tracking will be by a minimum of the applicable four (4) hours each and will zero every three (3) months with seniority being the deciding factor to start.
- 7. The applicable overtime ranking list for each job classification will be published and posted in all plant areas maintenance, laboratories, steam generation, logistics and the appropriate production buildings. The ranking, by employee, will include all overtime in each category worked up to and including Saturday of the previous week. The number of hours, shifts and call backs will not be on the list.

### **Scheduled Overtime**

- 1. All scheduled overtime, as defined in Article 5.02(b) and planned and posted by Supervision one week or more in advance, will be posted stating the work to be done. Where applicable, (i.e. maintenance, logistics, etc.) employees may sign up for a partial shift and the hours worked will be recorded as such.
- 2. Any employee desiring the posted overtime must sign up for the posted overtime by 7:00am Thursday. Note: When an employee does not work the approved overtime shift or works less than the full number of hours, either eight or twelve, it will be recorded as having worked the regular hours of the

- approved shift. This does not apply when the entire shift is shared by two or more employees (split shift).
- 3. Supervision will award the overtime by 5:30 pm of that Thursday by clearly noting and signing on the posting or issuing an overtime schedule as to who is assigned to work.
- **4.** Supervision will award the overtime based on qualifications required for the work, the ranking position, area (building) assignment, and subject to maximum hours or days legislation.

### **Continuous Overtime**

- Continuous overtime as defined in Article 5.02(c) and/or where overtime is required on short notice that cannot be posted will be offered by Supervision first to the employee with the least amount of overtime on the applicable time and one-half or double time list. Failing acceptance, Supervision will follow the provisions in Paragraph 4 of Scheduled Overtime.
- 2. When continuous overtime is required in conjunction with normal hours, the overtime will be awarded to the employee performing the work first. Failing acceptance, Supervision will follow the provisions of Paragraph 4 of Scheduled Overtime.

#### Call-Back Overtime

- Subject to the provisions of Article 5.02(a) and Continuous Overtime above, Supervision will offer call-back overtime to an employee following the provisions of Paragraph 4 of Scheduled Overtime.
- 2. Call-back overtime applies to any overtime, other than Scheduled and Continuous, where Supervision is required to call an off-site employee to offer overtime work. This does not apply when full shift coverage is required for 12-hour or 8-hour employees.

### Mandatory Maintenance Off-Hours Coverage

 The following maintenance trades are required to provide ongoing weekend coverage: Electricians, Instrumentation Technologists/Mechanics, Mechanics and Pipefitters subject to allowances for personal or family emergencies, illnesses or other unforeseen situations.

- 2. Mandatory coverage is for sixty-four (64) consecutive hours starting at 15:00 hours on Friday until 07:00 hours the next Monday morning. Should a designated holiday occur on the Friday or Monday of the coverage period, the period will be extended for twenty-four (24) hours.
- 3. The coverage will be scheduled on an annual basis and is subject to change to accommodate the work/ life balance of the employee through the change and/or exchange of hours using the application form provided by the Company.
- 4. The employee providing the mandatory coverage will receive \$4.00 per hour for the coverage period in Paragraph 2 as well as a minimum overtime pay at the applicable rate of four (4) hours for each call-back complied with as per Article 5.02.
- Coverage will be provided according to the following schedule:
  - a) Electricians one weekend each out of every eight weekends.
  - Instrumentation Technologists/Mechanics one weekend out of every five weekends.
  - Mechanics one weekend each out of every six weekends.
  - d) Pipefitters one weekend out of every seven weekends.

This schedule is subject to change where there may be an increase or decrease in the number of employees in the trade available. All overtime will be offered to an employee first before a contract employee or other is utilized on a full or part weekend basis.

- **6.** Employees will be given time off in lieu of time worked between 12:00 midnight and 07:00 during week nights.
- When an employee is unavailable for mandatory coverage due to illness, injury, vacation, bereavement or any other legitimate absence, it is Super-

vision's responsibility to find appropriate alternate coverage.

### **Telephone Protocol**

- 1. Let the phone ring a minimum of six (6) times.
- 2. Voice mail response leave message and proceed to next employee.
- 3. No answer proceed to next employee.
- **4.** Answered provide appropriate information Refused proceed to next employee.
- **5.** Answered provide appropriate information Accepted acknowledge.
- 5.07 Overtime records shall be kept by the Company and these records shall be available for review by employees in their buildings. Overtime records will be posted in all buildings on a weekly basis.

The Company shall continue its practice of providing the Union with weekly reports of hours worked by employees for each work week.

### ARTICLE 6 SHIFT PREMIUMS

- **6.01** (a) Employees on the twelve (12) hour shift schedule will receive 90 cents (\$.90) per hour for hours worked between 17:30 and 05:30. No premium will apply for hours worked between 05:30 and 17:30.
  - (b) Employees on the eight (8) hour shift schedule will be paid as follows for all shifts forming part of his regular schedule:

A rate of one and a half times the individual employee's regular straight time rate will be paid for hours worked on Saturday.

A rate of two times the individual employee's regular straight time rate for hours worked on Sunday.

(c) Employees on the twelve (12) hour shift schedule will be paid as follow for all shifts forming part of his regular schedule:

A rate of one and a half times the individual employee's regular straight time rate will be paid for hours worked 17:30 Saturday to 05:30 Sunday. A rate of two times the individual employee's regular straight time rate for all shifts commencing at 05:30 and 17:30 on Sunday.

- **6.02** In the event of an employee being called in to work on a shift other than his regular shift, the shift premium in 6.01 (a) for the "call in" shift shall apply.
- 6.03 In the event of a shift worker being scheduled to work prior to or beyond his regular shift, the shift premium in 6.01 (a) applicable to the shift worked prior to or beyond his regular shift shall apply.
- **6.04** Overtime shall not apply to the rate of the shift premium in 6.01 (a).

## ARTICLE 7 DESIGNATED HOLIDAYS & FLOATING HOLIDAYS

7.01 (a) The Company will pay all qualified employees for the following nine (9) Designated Holidays: New Year's Day, Good Friday, Victoria Day, Canada Day, Civic Holiday, Labour Day, Thanksgiving Day, Christmas Day, Boxing Day.

Payment for these days will be at the employee's current straight time hourly rate, exclusive of all types of premium payments.

Employees on the twelve (12) hour shift schedule will be paid for twelve (12) hours straight time pay and for those on the eight (8) hour schedule will be paid for eight (8) hours straight time pay.

Subject to Article 7.01 (c), an employee must work the full shifts immediately before and following the Designated Holiday to qualify, unless permission is granted by the Employee's Supervisor.

An employee who arrives for work more than one hour after the start of a qualifying shift must report to his department supervisor, or to the shift supervisor in their absence, before commencing work. If permitted to work the remainder of the shift, he will receive payment for the holiday.

(b) The Company will pay all qualified employees for three (3) floater holidays. The floating holiday will be taken at a time mutually acceptable to the employee in each calendar year. They may be taken individually or grouped together in two or more days.

Employees on the twelve (12) hour shift schedule will be paid for twelve (12) hours straight time pay and for those on the eight (8) hour schedule will be paid for eight (8) hours straight time pay.

To qualify for the three (3) floating holidays, an employee must have completed the probationary period. In the first and final calendar year of employment, an employee will have earned the floating holidays on the basis of one (1) day for each four (4) month period completed in the year.

Floater holidays may not be carried over to the next calendar year.

- (c) When an employee is on any approved absence, including illness, injury, lay-off or other approved leave, he will receive his holiday pay for any holiday which falls during the first two weeks of the absence, provided he worked his full shift prior to beginning the absence, unless permission for the absence is granted by his Supervisor.
- (d) Notwithstanding 7.01 (c), when a designated holiday occurs during the period an employee is receiving STD benefits, the employee will be paid the applicable designated holiday pay but not the STD benefit for that day.
- 7.02 A Designated Holiday may be observed on other than the calendar date by mutual agreement between the parties.
- 7.03 The hours constituting a Designated Holiday for the purpose of pay calculation for the individual employee will be the 24 hour period from the starting time of normal hours of work on the date that the holiday is observed through to the corresponding time on the following day.
- 7.04 When a Designated Holiday is observed on Friday or Saturday, an employee who has worked his qualifying shift preceding the holiday will receive payment for the holiday in that pay period. If the employee does not qualify for payment for the holiday, the amount received as payment for the holiday will be deducted from his pay.

- 7.05 If an employee is absent on one or both of the qualifying shifts for Christmas Day and Boxing Day, the maximum penalty to be imposed shall be the loss of one (1) of the paid days.
- **7.06** Employees who work on the Designated Holiday shall be paid a rate of two times the individual employee's regular straight time rate as follows:
  - (a) Employees on the eight (8) hour shift schedule for all hours worked on the Designated Holiday. Such payment at double time rate is in addition to payment for the holiday, if qualified.
  - (b) Employees on the twelve (12) hour shift schedule will be paid for shifts commencing at 05:30 and 17:30 on Designated Holidays. Such payment at double time rate is in addition to payment for the holiday, if qualified.
- 7.07 When the Company requires production facilities to be operated on designated holidays, the Company will post the required notice on the bulletin board in the building(s) affected a minimum of two weeks prior of the designated holiday(s). The employees scheduled to work the holiday will be required to work the shift unless a qualified replacement is found as per the overtime procedures. Employees scheduled to work the holiday with the highest seniority will be given the first opportunity to work or have the day off should a qualified replacement be obtained according to the normal double time overtime procedure.

Qualified by training for an employee is defined as being trained on the equipment and/or process to the satisfaction of the Process Engineer as signified by the training records.

The Company will make every effort to ensure all employees who would like the day off are given the opportunity. If an employee does work the holiday they will be given the option of taking another day off in lieu to be taken within the next three (3) months at a time to be agreed between the employee and his Supervisor.

Where a Holiday falls on a day that is not the Employee's regular work day, he shall receive the Holiday Pay for that day.

When an Employee agrees or is scheduled to work a Designated Holiday pursuant to Article 7.07, and he chooses to

request a lieu day, he shall submit the "Request for Vacation/ Floaters" form a minimum of one week prior to the Holiday. Where the Employee then works the Designated Holiday, he shall receive his double time pay for those hours worked in accordance with Article 7.06. If the Employee, having worked on the Designated Holiday, has elected a lieu day, under Article 7.07, the qualifier days, as described in 7.01 (a), shall then apply to the days before and after the lieu day. If an Employee is scheduled or has agreed to work on the Designated Holiday and has elected the option of a lieu day, and then fails to work on the Designated Holiday, he will not earn any wages for that day, and will not earn a lieu day or Holiday Pay.

## ARTICLE 8 TRANSFERS & JOB POSTINGS

- **8.01** For the purpose of this agreement, job transfers will be divided into three (3) categories:
  - (a) temporary loans on loan by the Company (whether within or outside his job classification) for a period of six (6) months or less.
    - In the case of a temporary loan expected to last six (6) months, the Company agrees to meet with the Union Executive to resolve the issues surrounding the affected process(es) and employee(s) involved in the temporary assignments required.
  - (b) temporary job transfers transferred (whether within or outside his job classification) following the job posting procedure for a period of more than six (6) months but not more than one (1) year.
  - (c) permanent job transfers transferred (whether within or outside his job classification) following the job posting procedure for a period in excess of one (1) calendar year.

### 8.02 Job Posting Procedure

- (a) All temporary and permanent job openings will be posted individually. Postings will include, as a minimum, the following information:
  - (i) the job classification,

- (ii) the department or production process building of the job opening,
- (iii) the qualifications required,
- (iv) whether the job opening is temporary or permanent and.
- (v) the deadline date of the posting.

Postings will be placed on the main bulletin board in each building in the plant and Bonnie Crescent Warehouse for a minimum of ten (10) calendar days.

Each employee absent from work during the original ten (10) calendar day posting prior to the final selection may make application to the specific job posting.

(b) Application forms will be available from the Human Resources Department and must be submitted to the Human Resources Department on or before the posted deadline. The Company will furnish a list of the applicants of each job opening to the Union at the deadline time.

### 8.03 Selection Process

- (a) From the applicants if any, the Company will select the successful applicant within twenty-one (21) working days of the expiration of the posting. All applicants will be acknowledged, however, only those with the requisite Qualifications may be interviewed.
- (b) The Employee with the greatest Qualifications shall be awarded the position, subject to the following. Qualifications as determined by the Company shall be on the basis of such terms as, skill, ability, process knowledge, experience, efficiency, and education required for the position. Where more than one applicants' Qualifications are considered equal, seniority will govern.
- (c) The Company will provide notice to the successful applicant, the Union and post the result on the main bulletin boards of each building in the plant and Bonnie Crescent Warehouse.
- (d) Upon request, the Company will provide the reasons for the decision to the unsuccessful applicant or applicants.

- (e) The successful applicant will be transferred as soon as possible to the new job from the date of selection and will be paid the hourly rate of that job on the start date of the transfer.
  - However, if the employee is not transferred within seventy five (75) calendar days from the date of being awarded the job, he shall be paid at the higher of his existing rate, or the rate of the new position, beginning at seventy five (75).
- (f) In the event that an employee transferring into a new position under this article creates a second vacancy, the Company's obligation to post shall be limited to the second vacancy.
- (g) Upon being placed in the position, the employee and Company shall have thirty (30) calendar days as a trial period, during which time either of them may elect for the employee to return to his previous position. This may be extended upon agreement of the Parties.

### **8.04** Temporary Loans

- (a) The Company may loan an employee as set out in Article 8.01 provided the employee has the required qualifications to perform the job, or can be adequately instructed to perform the job, in a timely fashion.
- (b) The employee with the highest seniority and required qualifications will be given the first opportunity to work or decline the temporary loan.
- (c) If necessary the employee with the lowest seniority and required qualifications will be loaned and thereafter successively until the required number of employees have been loaned.
- (d) An employee with seniority rights on loan will be paid the higher of the two hourly rates, if different.
- **8.05** Employees who transfer to a new job are not eligible to apply to another job posting for a period of six (6) months from the date of being awarded the new position.
- **8.06** Employees who are on a temporary loan will continue to accumulate seniority in accordance with Article 11.

8.07 An employee transferred to a position outside the bargaining unit, but within this plant, will retain the seniority credited to him at the time of transfer. Such an employee has forfeited his right of return to the bargaining unit except as a new employee.

## ARTICLE 9 VACATIONS

- 9.01 For the purpose of calculating vacation entitlement, the increments shall be expressed in days of vacation. Employees will be eligible for vacation and vacation pay in the current calendar year on the following basis:
  - (a) An employee with less than 1 year of seniority as of June 30th will receive vacation pay calculated at a vacation rate of 4%, as described in Article 9.05. Time off for vacation will be on the basis of four (4) days for twelve hour shift employees and five (5) days for eight hour shift employees for six completed months of seniority as of June 30th and prorated for completed months of seniority above or below that figure.
  - (b) An employee with one (1) year of seniority but less than five (5) years as of June 30th will receive vacation pay calculated at a vacation rate of 4% as described in Article 9.05. Time off for vacation will be on the basis of eight (8) days for twelve hour shift employees and ten (10) days for eight hour shift employees.
  - (c) An employee with five (5) years of seniority but less than twelve (12) years as of June 30th will receive vacation pay calculated at a vacation rate of 6% as described in Article 9.05. Time off for vacation will be on the basis of twelve (12) days for twelve hour shift employees and fifteen (15) days for eight hour shift employees.
  - (d) An employee with twelve (12) years of seniority but less than twenty (20) as of June 30<sup>th</sup> will receive vacation pay calculated at a vacation rate of 8% as described in Article 9.05. Time off for vacation will be on the basis of sixteen (16) days for twelve hour shift employees and twenty (20) days for eight hour shift employees.

- (e) An employee with twenty (20) years of seniority but less than twenty-five (25) as of June 30th will receive vacation pay calculated at a vacation rate of 10% as described in Article 9.05. Time off for vacation will be on the basis of twenty (20) days for twelve hour shift employees and twenty-five (25) days for eight hour shift employees.
- (f) An employee with twenty-five (25) or more years of seniority as of June 30th will receive vacation pay calculated at a vacation rate of 12% as described in Article 9.05. Time off for vacation will be on the basis of twenty four (24) days for twelve hour shift employees and thirty (30) days for eight hour shift employees.
- (g) An employee who accumulates and is credited in the current calendar year with five (5), twelve (12), twenty (20), or twenty-five (25) years of seniority between June 30<sup>th</sup> and December 31<sup>st</sup> shall receive an additional four (4) or five (5) days, as appropriate, of vacation. Vacation pay will be accumulated as an additional two percent (2%) on top of normal vacation rate.

Should an employee work both the twelve (12) hour shift and the eight (8) hour shift during a calendar year, vacation time will be prorated based on the ratio of twelve (12) hour shift time versus eight (8) hour shift time.

- 9.02 Where a Designated Holiday falls during an employee's scheduled vacation, the day shall count as the Designated Holiday and not be a day drawn from the employee's vacation days.
- 9.03 All vacation must be taken during the calendar year at a time mutually agreed upon between the employee and the Company. Employees may request vacations starting October 1st until December 31st for the following year's vacation. These requests will be confirmed on or before the following January 15th. Conflicts will be resolved by seniority.

Subsequent vacation requests during the year will be based on a first come, first served basis. For subsequent vacation requests, Overtime Procedures will be used in an effort to find a replacement to grant that vacation request. Where overtime is required to cover for vacation periods, all such overtime will be posted for each vacation request. Vacation requests must be made a minimum of seven (7) days before the desired vacation period.

The Company will endeavour to give three (3) months notice and grant at least eight (8) vacation days for 12-hour eligible employees or ten (10) vacation days for 8-hour eligible employees during the summer months. If necessary to meet emergency requirements, payment of vacation pay may be made for days 17 through 24 for 12-hour employees or days 21 through 30 for 8-hour employees, without time off being granted. At the option of the employee, payment of vacation pay may be made for days 17 through 24 for 12-hour employees or days 21 through 30 for 8-hour employees, without time-off being granted providing the employee was on an approved leave of absence for four (4) weeks or more.

It is understood that, with mutual consent, payment of vacation pay may be made without time off being granted for days 9 through 24 for 12-hour employees or days 11 through 30 for 8-hour employees.

9.04 When an employee is laid off for an indefinite period, or employment terminated, the employee will be paid within seven (7) calendar days from the date of lay-off or termination, the amount of his vacation pay accrual to the date of lay-off or termination, which shall be either 4%, 6%, 8%, 10% or 12% according to seniority qualifications as set out in Article 9.01. When the laid-off or terminated employee is entitled to pay in lieu of notice under the *Employment Standards Act*, the above accrual shall include his applicable vacation pay for the period of notice paid in lieu.

When an employee is on an approved absence during which the Employer provides payment of a monetary benefit to the employee, including those set out in the Employment Standards Act, as may be applicable, or Article 3.03 of the Group Life and Health Insurance Plan, the amount of benefits paid will be included in the employee's current weeks earnings for that period as set out in Article 9.05.

9.05 (a) Vacation pay will accrue to each employee's account on a pay period basis and recorded on the pay information slip. The amount accrued will be calculated as the percentage set out in 9.01 of the current pay period earnings. The percentage will be based on the seniority

which would be accumulated and credited by the end of the calendar year in which payment is to be made, (see 9.05(b)) assuming continuous employment. If the employee ceases to be actively employed or if employment is terminated for any reason prior to the qualification date, any vacation pay paid in advance, in anticipation of attaining qualification, shall be deducted from the final pay.

- (b) Vacation pay will be paid out twice per year on a pay slip separate from regular pay period earnings. The first payment will be made not later than the second regular pay date in January. The second will be made not later than the second regular pay date in July. The amount of each payment will be the total amount of vacation pay accrued during the preceding six months.
  - I.E: January payment will be based on accruals during the six months July 1 to December 31. July payment will be based on accruals during the six months January 1 to June 30.

An employee on Pregnancy, Parental or Family Medical Leave may elect to defer payment of their vacation pay until they return to work.

- **9.06** The vacation pay calculated in accordance with Article 9, will be a minimum amount of \$500.00 for the year or the two payments in January and July of \$250.00 for each week of vacation entitlement when the employee:
  - Is a regular, full time employee, with one (1) or more years of seniority, and
  - Has been unable to work for a portion of the six months preceding December 31 or June 30, as applicable because of verified illness or injury, and
  - Has accrued vacation pay during the previous six months which is less than the minimum amount stated above, and
  - Has worked three or more months during the twelve months ending December 31 or June 30 as applicable.

### ARTICLE 10 SAFETY AND HEALTH

- 10.01 The Company shall continue to provide for the safety and health of its employees by providing necessary safety devices, safety equipment and protective clothing that the Company considers necessary.
- 10.02 Where the Company and Union agree that it is necessary for safety reasons to stipulate that at least two employees should work in a particular area and/or process, it shall be included in the operating procedures. As new operations are added or changes are made to existing processes, the Joint Health & Safety Committee will review these operations/processes and make recommendations pertaining to foregoing requirement.

At least two employees must be present during the high pressure operating steps of the following processes: Bldg 18 (i.e. Pilot Plant) high pressure equipment; Bldg. 33 PAO hydrogenator; and Bldg. 37 BLE Autoclave.

In the event there is only one operator working on a process, the operating procedures will stipulate measures required to ensure the safety of that individual. These measures will include:

- Process controls (e.g. operating limits).
- Procedures for safely shutting down the process or placing it in a safe stand-by state.
- Restricting certain tasks, where applicable.
- Operating the process at a reduced or limited rate, where applicable.

Where there is only one operator working in a process building, a means for the operator to contact, or be contacted by, another person at the site (e.g. radio), will be provided.

- 10.03 The Company and Union recognize the importance of Health and Safety of all employees and the Community. All employees will follow the provisions of the *Occupational Health and Safety Act* and Regulations, as amended, and Company policies and programs.
- **10.04** An employee shall report promptly any injury, no matter how minor, to his supervisor.

If, through an injury at work, it is found necessary to remove the injured employee from the plant, he shall be paid for his entire shift that day.

If the injury requires medical attention at a health care facility outside the plant, the employee will be paid, at the appropriate rate, for time spent receiving the initial medical care, travel time, and travel expenses incurred from and to the place of employment or to home, as may be applicable.

10.05 (a) A Joint Health and Safety Committee will be comprised of five (5) employee members appointed by the bargaining unit and up to five (5) management members appointed by the Company. There shall be two (2) co-chairpersons, one from the Company and one from the Union who shall alternate the chair at meetings.

The functions of this committee will be in accordance with the Occupational Health and Safety Act as amended.

Meetings of the Committee will be scheduled by the co-chairpersons. An employee appointed as a member by the Local Union will be paid at their applicable hourly rate for time spent at such meetings.

- (b) The Company will, upon request, co-operate in furnishing available information to the Committee on materials and substances used in the plant.
  - The Union agrees to respect the confidential nature of information pertaining to materials and manufacturing processes made available to the Committee under this Article.
- (c) The Company will provide paid time and cover the training costs for certification, and refresher training after every three years, for worker members serving on the Joint Health and Safety Committee.
- (d) The parties agree that the Joint Health and Safety Committee should post the minutes of the Committee's meetings within one week of the meeting. A copy of the minutes shall be sent to the Unit Chair, and the Plant Manufacturing Manager, or their designates.
- 10.06 An employee will not work alone on a chemical process and/ or equipment until trained according to the training policy.

### ARTICLE 11 SENIORITY

11.01 (a) The Company values and appreciates faithful service of employees and will endeavor to provide continuity of employment and preference in advancement. It is recognized that the safe and efficient operation of the plant is of first importance for the prosperity of the employees and the growth and survival of the Company.

On qualifying for seniority, it will be expressed in terms of time of a continuous nature and will include the probationary period, temporary employment period, time on an approved leave of absence as provide in Article 11:04, and time on lay-off up to six (6) months for an employee with less than five (5) years of seniority and up to one (1) year for an employee with five (5) or more years of seniority as of the date of lay-off.

Seniority accumulated while on an approved leave of absence or lay-off will not be credited until the employee returns to work.

Seniority rights will commence upon completion of the probationary period as described in Article 11.02.

- (b) The Company will furnish the Recording Secretary of the Union, upon request, but not more than every six (6) months, with a list of employees in the bargaining unit in order of their plant seniority and a separate list of the current addresses of the employees in the bargaining unit.
- (c) The Company will provide, in advance, written notice to the Union for all temporary employees the Company wishes to hire, recruited through the HR department or by HR through a temporary staffing agency, in the bargaining unit to temporarily replace absent employees. The notice shall include the particulars regarding the temporary employee such as the purpose of the hire, the job the employee will be performing and the expected duration of the period of employment. Temporary employees covered in this section will receive no seniority rights but will receive remuneration based on the job classification and rates in Article 16 and pay the applicable Union dues as provided in Article 3.

- (d) The Company will provide, within 30 days of the end of each quarter, a record to the Union of all outside maintenance contractor employees the Company hired that quarter, to perform work normally performed by bargaining unit employees. The record will state the number of hours worked, the job classification and company name. Outside contractor employees receive no seniority rights. The Company will pay to the Union, as provided in Article 3, the appropriate Union dues based on the job classification hourly rate of pay as listed in Article 16.
- **11.02** An employee is a probationary employee until completion of six (6) months' continuous employment, and has no seniority rights during that period.

The only exception to the foregoing is that an employee laid off after completing at least three (3) months' continuous employment and who is re-employed within six (6) months of the date of lay-off will, upon completion of three (3) additional months' continuous employment, be considered to have completed the above probationary period and be credited with actual time worked during the two periods, thus qualifying for seniority rights.

### 11.03 Lay-offs

In the event a lay-off becomes necessary, the following procedure will apply:

- (a) Prior to any lay-off, the Company and the Union representatives will meet to discuss such lay-off.
- (b) All probationary, temporary, contract and/or outside contractor employees performing bargaining unit work will be placed on lay-off before employees with seniority rights are placed on lay-off. If the necessary skill set is not available among the employees to be laid off the above may be retained.
- (c) Where it is the deciding factor in a lay-off, progression or transfer, seniority will be confined to three departments: Trades Department, Laboratory Technician Department and Production Department. The three departments are further described in Article 11.05.

Seniority accumulated in the Trades Department, Laboratory Technicians Department or Production Depart-

ment will not be recognized in the other departments. Seniority will accumulate in one department as department seniority but all seniority will accumulate as plant seniority.

All lay-offs will be based on Department Seniority except in the Trades Department where lay-offs will be based on job classification.

The employee with the least seniority in the department or, in the Trades Department, job classification, will be placed on lay-off first and continuing in succession until the required number are placed on lay-off.

When employment start dates are identical, any lay-off will start with the highest employee number.

For further clarity, when it becomes necessary to layoff or transfer an employee or employees of a job classification in the Trades Department where a class of qualification or certificate level applies, the lay-off or transfer will, commencing with the lower rated class or level, be in order of seniority within each class of qualification in the Maintenance Department and within each certificate level in the Steam Generation, Groundwater/Wastewater Treatment Department.

- (d) Employees with seniority rights have the right of recall for up to one (1) year.
- (e) Employees on lay-off will be recalled in reverse order of lay-off. Where practical, no new, temporary, contract and/or outside contractor employees will be hired to perform bargaining unit work until all employees with right of recall and the necessary skill set to perform the required work have been recalled.
- (f) Employees on lay-off are responsible to inform the Company of their current address and telephone number.
- (g) An employee on lay-off, will, upon receipt of recall, report to the Company within two (2) weeks after receiving such notice. To mitigate any dispute, the dated postmark of the written recall notice will determine the start date of the two (2) week period. Failure to report within the two-week period will result in loss of further employment rights with the Company.

- (h) Department seniority will terminate upon lay-off and must be re-established on return to work.
- (i) The company shall conduct a job posting for a required vacant position created by a layoff, in accordance with Article 8, with the obligation to conduct posting only for this first such vacancy.
- (j) Except as provided in the paragraph immediately following, an employee with three or more years of plant seniority who is to be laid off may, providing he has greater seniority and can qualify in a two week period, replace the employee with the least seniority in the Trade Department, Laboratory Technician Department or in the Production Department, as applicable.

The foregoing provision will not apply if the employee to be replaced is enrolled in a recognized apprenticeship program in the Trades or Production Departments or enrolled in the modular training program for Operating Engineers.

(k) In the event of a lay-off involving employees with seniority, the Company will give such employees a minimum of fourteen (14) days notice of lay-off. When an emergency shutdown occurs, the Union and the Company will meet to arrange shorter notice. In case of probationary employees, shorter advance notice of lay-off may be given. In any event, the Company will follow, as a minimum, the provisions of the current Ontario Employment Standards Act.

Seniority, for a maximum of one (1) year, accumulated while on lay-off will not be credited until the employee returns to work.

### 11.04 Leave of Absence

An employee with seniority who desires a leave of absence will submit an application to the Department Head and Human Resources Manager. A leave of absence may be granted without loss of seniority subject to the following conditions:

(a) Leave of absence for a specified period of time not in excess of 3 months and subject, upon application by the employee, to not more than 3 extensions for a total period not in excess of one year, may be granted to an

employee due to personal non-occupational injury or illness.

An application by the employee for an extension beyond the one-year period will be reviewed by the Company, the Union and the employee concerned and may, at the option of the Company, be granted if it is indicated that the employee will be able to return upon expiration of such extension; otherwise, an extension will not be granted beyond the one-year period.

Seniority will be accumulated during a leave of absence, under the above conditions, up to an amount equivalent to the seniority credited to the employee at the start of such leave but not in excess of six (6) months accumulation for an employee with less than 5 years' seniority, and not in excess of one (1) year for an employee with 5 or more years' seniority.

- (b) Leave of absence for a specified period of time not in excess of 1 month and not subject to any extension may, at the option of the Company, be granted upon written application by an employee due to a serious personal necessity or reason. If granted, seniority will be accumulated for the period of such leave.
- (c) A member of the Union elected or selected for union duties with the United Steelworkers, which necessitates his absence from employment with the Company may, upon written application by the Union, be granted leave of absence for a specified period of time up to one year.

If leave is granted for this purpose seniority status will be retained, and on return to work, accumulation of seniority will be credited for the period of the leave.

(d) Subject to physical qualifications and seniority status, an employee returning from an approved leave of absence of up to one year shall, within one week of notifying the Company of his desire to return, be returned to the job on which he was employed immediately prior to the leave of absence.

If the leave of absence has extended beyond one year, or physical capabilities or seniority status do not permit

- return to his former job, he will be placed on an available job consistent with his physical capability and seniority status.
- (e) An employee on leave of absence due to a compensable injury incurred while in the employ of the Company is not subject to the limitation on accumulation of seniority provided in this Article. He may return to work when medically able to do so, and, if unable to perform his previous job, may be placed on an available job consistent with his capabilities and seniority status, and will be credited with full accumulation of seniority for the period of the leave.

#### 11.05 Department Seniority Rights

- (a) The following are recognized as departments for purpose of this agreement:
  - 1. Trades Department will consist of Maintenance, Steam Generation and Groundwater/Wastewater Treatment (generally buildings 20, 45, 44, 54, 62 and groundwater wells).
  - Laboratory Technicians Department will consist of the Quality Assurance Lab, Factory Services and, Environmental Department.
  - **3.** Production Department will consist of Production Process Buildings:
    - Buildings numbered 13, 14, 15 including Tank Farm, 18,19, 25, 28, 33 including Tank Farm, 36 and 37 including Main Tank Farm and 37 Tank Farm.
    - (ii) Logistics Bldg 7 and Bonnie Crescent Warehouse

Exception to the above is the Vitaflo operations that will include building 28 and 36 and be considered a single Production Process Building.

- (b) Department Seniority is based on the following criteria:
  - An employee with department or Production Process Building seniority transferred by the Company to a different Department or Produc-

tion Process Building will retain seniority in the Department from which he was transferred until employed in another Department or Production Process Building for ninety (90) days. Seniority from the initial date of transfer, or total production seniority in case of transfer or transfers between Production Process Buildings, will then be credited in the new Department or Production Process Building and Seniority in the previous Department or Production Process Building terminated.

- 2. An employee who transfers from a Department at his own request terminates seniority in that Department or Production Process Building immediately on transfer. Seniority from the date of transfer, or total production seniority where the transfer is between Production Process Buildings will be credited as Production Process Building Seniority when employed in the new Department or Production Process Building for ninety (90) consecutive days.
- 3. Upon three (3) consecutive months of employment in any department, department seniority rights shall apply for an employee with seniority rights as provided in (2) above. For the Production Department, each Production Process Building will be recognized as distinct for the purpose of accumulating department seniority. For probationary employees five (5) consecutive months of employment shall apply except as provided in the second paragraph of Article 11.02.
- **4.** Overtime ranking as provided by Article 5 is excluded from the time period above for employees with seniority rights.
- 5. Total plant seniority will be credited to the employee as department or production process building seniority upon completion of three (3) months of employment in the department or production process building.
- **6.** For an employee as described in 11.05(b) 2, employment time spent outside of the department due

to a temporary loan as described in Article 8 will be included in the department from which he was loaned.

- 7. Department or production process building seniority will terminate on the date that an employee transfers from the department or production process building at the employee's request through the job posting procedure as provided in Article 8.
- 8. "Duty To Accommodate" Notwithstanding any other provisions of this Collective Agreement, the Company and the Union agree that all employees who can no longer perform the essential duties of their job will be accommodated in accordance with the Ontario *Human Rights Code*. The Company and the Union will, meet to review the availability of work that is suitable in consideration of the employee's disability and/or restrictions. Such accommodation is not necessarily restricted to duties of only one job classification and if no suitable job is available in the bargaining unit, consideration will be given to jobs outside the bargaining unit.

#### 11.06 Termination and Loss of Seniority Rights

A complete loss of seniority rights and cessation of employment will occur for any of the following reasons:

- (1) When an employee voluntarily resigns or quits employment with the Company. An employee who resigns or quits employment will notify the Company seven (7) days before separation.
- (2) Discharge for cause. However, should an arbitrator or other relevant authority overturn the discharge, all time away from employment will be credited to the employee upon return to work.
- (3) Failure to return to work following recall from lay-off as provided in Article 11.03(g) or from an approved leave of absence as provided in Article 11.04.
- (4) Upon expiration of recall rights in accordance with 11.03(d).
- (5) Utilizes a Leave of absence for a purpose other than for which it was granted.

(6) Absent from three or more consecutive scheduled shifts without contacting the Company during such time and providing a satisfactory explanation, unless unable to contact the Company.

#### ARTICLE 12 GRIEVANCE PROCEDURE

#### 12.01 Grievance Procedure

The Company and the Union agree to meet promptly to discuss and negotiate any complaints or grievances that may arise, and make every effort to settle any matter at issue as quickly as possible. The following procedure will be followed progressively for the resolution of grievances and disputes.

Nothing contained herein shall prohibit an employee from presenting any personal matter directly to the Management.

Grievances related to suspensions and/or terminations shall commence with Step 3 of the grievance procedure and will be reduced to writing and submitted to the Human Resources Manager within three (3) working days.

#### Step 1

No grievance shall be considered if the circumstances occur or originate more than fourteen (14) working days prior to filing the grievance at step one. A meeting will be held between the aggrieved employee and the steward and/or chief steward and the first line supervisor or department head in his absence. A response from the supervisor or department head will be provided within seven (7) working days of the Step 1 meeting.

#### Step 2

Failing settlement at Step 1, the grievance will be reduced to writing and submitted to the Human Resources Manager, including provisions allegedly breached and brief statement of the facts relied upon, within seven (7) working days of the Step 1 response. A meeting of the Union Grievance Committee and management representatives will be arranged within fourteen (14) working days of receipt of the written grievance. A written response from the Company will be provided

to the Union Grievance Committee within seven (7) working days of the meeting.

#### Step 3

Failing settlement at Step 2, the Union may, within seven (7) working days of the receipt of the Step 2 response, request a meeting of the Union Grievance Committee with the Union President and management representatives including the Director of Manufacturing. This meeting will take place within fourteen (14) working days of receipt of the request for the meeting. A written decision from the Company will be provided within seven (7) working days from the date of the meeting.

#### Step 4

Failing settlement at Step 3, the grievance may be referred to a qualified mediator mutually agreed upon. The Company and the Union shall equally share the costs incurred by this process. Mediation shall be without prejudice to either party's position should the matter proceed to Arbitration. The referral to mediation shall not affect any timelines set out herein. Notice of a desire to mediate the grievance must be provided by one party to the other within seven (7) working days of the Step 3 reply, failing which the matter shall be deemed settled, or referred to arbitration within a further seven (7) working days, in accordance with the provisions in Article 13.

- 12.02 In the event that a response at any step of the above Grievance Procedure is not provided in the time lines specified it shall be deemed denied on the date the Response is due, and may be referred to the next Step in the Grievance Procedure. If it is not so referred to the next Step it shall be deemed abandoned. The time lines in any Step may be extended upon written mutual agreement of the parties.
- 12.03 Where there is reference to "working days" in the Grievance or Arbitration Process it shall mean a day other than Saturday, Sunday, or a Designated Holiday.

#### ARTICLE 13 ARBITRATION

13.01 Should an agreement not be reached through the foregoing steps of the grievance procedure, it shall be referred to Arbitration in accordance with the provisions herein, or as permitted under the Labour Relations Act. Any matter so referred to Arbitration shall be handled as follows:

Where a difference arises between the parties relating to the interpretation, application or administration of this Agreement, including any question as to whether a matter is arbitrable or where an allegation is made that this Agreement has been violated, either of the parties may, after exhausting any grievance procedure established by this Agreement, notify the other party in writing of its desire to submit the difference or allegation to arbitration providing such written request is made within fourteen (14) working days of completion of the final step of the grievance procedure. The notice shall contain three (3) proposed sole arbitrators that the referring Party suggest to hear the matter. The other party shall respond within five (5) working days, either accepting one of the arbitrators proposed, or with an alternate list of arbitrators for consideration. If the parties jointly agree the matter may be heard by a Board of Arbitration, in which case they shall inform of their nominees in the above timelines and the nominees shall then agree on a Chairperson within a further ten (10) working days. Failing agreement of the sole arbitrator or Chair, the parties may request an appointment pursuant to section 50 of the Labour Relations Act.

The Arbitrator shall hear and determine the difference or allegation and shall issue a decision and the decision shall be final and binding upon any employee and upon the parties affected by it.

Where a Board of Arbitration is substituted as provided above, all other applicable provisions of this Article, including the procedure for selection of the Board shall continue to apply.

13.02 The Company and the Union may within three (3) regular working days from the date of appointment of the Arbitrator, sign a joint stipulation of the dispute on the question which is to be arbitrated. The Arbitrator shall not have jurisdiction

to alter or change any of the provisions of this Agreement, and such jurisdiction shall be limited to a decision on the dispute in question set forth in the stipulation. All decisions of the Board of Arbitration arrived at in accordance with the provisions of this paragraph shall be final and binding on all parties concerned.

- 13.03 Where applicable, the decision of a majority shall be the decision of the Arbitration Board, but if there is no majority the decision of the chair shall govern.
- 13.04 Each of the parties hereto will bear the expenses of the nominee, if applicable, appointed by it. The parties will jointly bear the expense of the chair or sole Arbitrator, and other expenses incurred, if any.
- **13.05** It is understood and agreed that any of the time limits provided in Article 13 may be extended by mutual agreement of the parties in writing. Should the time limits elapse without an application for an extension, the grievance will be deemed abandoned or settled, as applicable.

#### ARTICLE 14 DISCIPLINE

14.01 The Company and the Union agree that in the issuance of discipline, the concept of progressive and corrective discipline shall prevail in the majority of cases. The Company agrees to issue discipline on an equitable basis as far as it is practicable.

The Company may, depending on the severity of the violation, for good and sufficient cause, skip the progressive discipline concept and issue discipline up to and including immediate termination.

The Company and the Union agree, subject to the Union's right to challenge any such discipline through the grievance procedure, that progressive discipline for suspensions shall be administered wherever appropriate in the following three levels:

(1) A Level 1 suspension shall be equivalent to a one-day suspension whether or not a twelve (12) hour or an eight (8) hour shift. Employees at Level 1 shall have a one-day suspension recorded as discipline issued.

- (2) A Level 2 suspension shall be equivalent to a 3-day suspension of 8 hours per day and therefore a person working 12 hour shifts shall lose two – 12 hour shifts to reach the three-day equivalency. Notwithstanding the foregoing, employees at Level 2 shall have a three-day suspension recorded as discipline issued.
- (3) A Level 3 suspension shall be equivalent to a 5-day suspension of 8 hours per day and therefore a person working 12 hour shifts shall lose three – 12 hour shifts to reach the 5 day equivalency. Notwithstanding the foregoing, employees at Level 3 shall have a 5-day suspension recorded as discipline issued.
- (4) A workday for which double time is paid, e.g. Sunday or a Designated Holiday, may be served in a Level 2 or 3 suspension and in which case shall be counted as 2 days of the suspension.
- (5) The Company shall not issue discipline by way of a Level 1 suspension which would cause an employee who is employed on a Sunday or a Designated Holiday as part of his regular workweek to lose pay for that specific day; moreover, the Company shall postpone the suspension to a straight time shift except where such suspension is required to be issued immediately for good and sufficient cause.
- (6) In the event the Company suspends an employee either directly, before, during or after the Designated Holiday, it will not affect the employee's entitlement to be paid for the holiday in accordance with the Employment Standards Act, provided the employee works his full regular shift scheduled before and after the suspension.
- (7) A record of any discipline given by the Company to an employee will be furnished to the Union at the time the discipline is administered.
- (8) When an employee is to be disciplined Union representation by means of a steward, chief steward or executive officer will be present.
- (9) When an employee is to be suspended or discharged, the chief steward, president or, in his absence, another executive officer must be notified and will be present at the meeting.

- (10) Any verbal or written discipline will be stricken from the employee's record after one (1) year from the date of issue providing no additional disciplinary action of like manner is recorded during the above time limit.
- (11) Any disciplinary suspension of three (3) days or less will be stricken from the employee's record after two (2) years from the date of issue providing no additional disciplinary action of like manner is recorded during the above time limit.

# ARTICLE 15 PRIVACY OF EMPLOYEE INFORMATION

15.01 The Company and Union agree to follow all applicable privacy legislation, statutes and regulations in regards to all employees' rights to privacy of their personal information and personal health information.

The Company respects the privacy rights and dignity of all employees. Although it is essential for the Company to gather and maintain certain information about its employees, it is the Company's policy to acquire and retain only the information that is required by law or necessary for effective corporate operation.

Only authorized Company employees with a valid work related reason may have access to and use of records kept by the Company containing employee personal information. All personal information matters relating to the employment or employeent history of any employee or former employee are confidential information. No Company official is authorized to disclose any such information to any other person, organization or institution, without the express consent of the employee, except as may be required or expressly permitted under applicable laws.

The parties further agree that, with respect to personal health information, as defined by the Personal Health Information Protection Act, 2004, as amended, they shall comply with the provisions of the Act regarding the collection, use and disclosure of such records.

An employee who believes his privacy rights were violated should immediately report the alleged violation to the Company's Manager of Human Resources for Canada and/

or the Bargaining Unit Committee. The Company and the Bargaining Unit Committee will work together to resolve the substance of the complaint or report.

None of the above prohibits an employee from resolving the matter individually or applying the provincial or federal legislated complaint procedures.

#### ARTICLE 16 JOB RATES AND PROMOTIONS

**16.01** If the Company establishes a new classification or reclassifies a position, the Union will be notified in writing and if requested, a meeting arranged for an explanation of the change. A grievance may be submitted in writing within seven (7) days of the notification or meeting and will be considered a grievance under Article 12, commencing at Step 2.

#### 16.02 (a) PRODUCTION

An employee, without previous experience, assigned on hiring to a production job, with the exception of logistics, will be required to register with the Ministry of Training, Colleges and Universities in the Chemical Process Operator Apprentice Program and complete all phases of the program. The employee will receive an increase of one quarter of the difference between the minimum starting rate and the full rate for each written phase of the program completed until he reaches the classification of the job.

## (b) STEAM GENERATION, WASTE WATER/GROUND WATER TREATMENT

An employee with seniority transferred to the classification of Operating Assistant or Expeditor will receive the classified rate of the job. This employee is required to be enrolled immediately in the Modular Training Program for Operating Engineers. They are required to write the first exam within the first six months in the position. Upon successful completion of this exam the Operating Assistant or Expeditor will receive an increase of the midpoint between the Operating Assistant or Expeditor rate and the corresponding fourth class rate, as applicable. All examinations necessary to obtain a Fourth Class Operating Engineer's

certificate must be completed within eighteen months as of the date of transfer, which may be extended to twenty four (24) months at the discretion of the Chief Operating Engineer in consultation with the Union. The Employer shall pay the cost of an examination upon successful completion, and where an employee requests study time during working hours, the Chief Operating Engineer will make every reasonable effort to grant such time, subject to operational requirements. If the employee has failed to obtain the required certificate within the above time limit, the employee may be replaced by another employee through the job posting procedure.

Employees qualifying or hired into any level of the Operating Engineer classifications are encouraged and expected to continue their training to the next level. Once obtaining the next level the employee will be paid at that rate once the certificate is obtained and presented to the Company.

#### (c) LABORATORY TECHNICIAN

An employee on being hired as a technician in one of the laboratories will be given the classified rate of a 2nd Class Technician.

On showing satisfactory performance, an additional minimum increase of 45 cents per hour will be given after each six-month period, and the classified rate of 1st Class Technician after eighteen (18) months of employment.

To achieve the classification of Senior Technician the employee must be performing job functions above that of the 1st Class technician and of a developmental nature. This classification will be limited to one employee in each technical area. A Technical Area is each distinct laboratory where there are bargaining unit employees performing work.

#### (d) LOGISTICS

An employee assigned on hiring or transfer through the job posting procedure in Article 8 to the classification of Helper-Materials Handler must obtain their forklift

and Transportation of Dangerous Goods certification within the first 60 days through training opportunities provided by the Company in this time period.

A new employee will receive, upon satisfactory performance, an equal monthly incremental pay increase over the probationary period until the full rate of Materials Handler is achieved.

An employee assigned thru the job posting procedure of Article 8 will receive, upon satisfactory performance, an equal monthly incremental increase over sixty (60) days until the full rate of Materials Handler is achieved.

## (e) REGULATED MAINTENANCE TRADES AND STEAM

Employees who transfer to the Trades Department must enroll in the applicable training program directly upon transfer to that department. Should the applicant fail to enroll and/or fail to complete the training program satisfactorily, the employee will be placed in the first vacancy within the plant for which he is qualified.

Completing the training program satisfactorily shall be deemed to mean obtaining the diploma, certificate or ticket in an appropriate timeframe as determined by the governing body of the organization responsible for conducting the training program. In the case of no governing body, the Company and the Union will set the Standards to qualify the employee.

This Apprenticeship Program provides an opportunity for employees to participate in a workplace based training system to acquire the skills necessary to be certified as a Regulated Trades under the Apprenticeship and Certification Act of the Province of Ontario.

The selection of apprentices, will be in accordance with the Job Posting Procedure in Article 8 of the Collective Labour Agreement.

Apprentices will be required to pass the standards set by the Ministry of Training, Colleges and Universities. The Company will also periodically review the on-the-job performance and training progress of the employee, which must be maintained at an acceptable level. Failure to do so will result in the employee's removal from the Apprenticeship Program and placement on a job in the bargaining unit for which the employee is qualified. Such transfer must not result in the displacement of any other employee from their regular job except, that if there is no vacant job for which they can qualify. The employee may displace the most junior employee in the plant if the transferring employee has more seniority and possesses the necessary qualifications and ability to perform the job.

Upon successful completion of each phase of training or block release segment of schooling at a Community College the following wage structure will apply:

Entry Level – 80% 1st Class Maintenance Trades Classification

Basic Level – 85% 1st Class Maintenance Trades Classification

Intermediate Level – 90% 1st Class Maintenance Trades Classification

Advanced Level – 95% 1st Class Maintenance Trades Classification

Certificate of — 100% 1st Class Maintenance Trades Classification

Certificate of Qualification 1st Class Maintenance Trades Classification

The Company will pay each apprentice a travel cost at

The Company will pay each apprentice a travel cost at the applicable Company rate and all books and learning material costs while they attend the block release segment of schooling at a community college.

When requested by the Company, apprentices will provide, in a timely fashion, all relevant information such as grades, attendance sheets, etc., pertaining to their participation in the program.

The Company and Union agree that all apprentices in the Regulated trades that are affected by the block release program will be reimbursed for their net loss of wages. This will be the difference between the current system of payment and the employee's straight time hourly rate for 40 hours per week times the number of weeks absent from work. The difference will be paid upon the successful completion of the program and the

obtaining of the Certificate of Qualification. The employee will be required to produce all documentation of Government payments and Income Tax information.

The Company will pay the cost for an employee to renew his trade ticket that is required for his position.

**16.03** Subject to provisions of Article 16.01, the following job classifications and rates shall apply under the terms of this Agreement:

#### MAXIMUM HOURLY CLASSIFICATION RATE

	HOURLY RATE EFFECTIVE:			
JOB CLASSIFICATION	2015	2016	2017	
PRODUCTION				
Utility Person	24.34	24.83	25.33	
Expediter	26.10	26.62	27.15	
Process Operator	24.23	24.71	25.20	

	HOURLY RATE EFFECTIVE:			
JOB CLASSIFICATION	2015	2016	2017	
LOGISTICS				
Lead Shipper	25.82	26.34	26.87	
Materials Handler	24.34	24.83	25.33	
Helper	22.02	22.46	22.91	

	HOURLY RATE EFFECTIVE:			
JOB CLASSIFICATION	2015	2016	2017	
TECHNICIANS				
Senior Technicians	26.22	26.74	27.27	
1st Class Technicians	25.82	26.34	26.87	
2nd Class Technician	24.23	24.71	25.20	

	HOURLY RATE EFFECTIVE:				
JOB CLASSIFICATION	2015	2016	2017		
TRADES					
STEAM GENERATION/WASTE W	/ATER/GI	ROUNDW	/ATER		
Assistant Chief Operating Engineer	31.29	31.92	32.56		
Second Class Operating Engineer	30.75	31.37	32.00		
Third Class Operating Engineer	28.24	28.80	29.38		
Fourth Class Operating Engineer	24.99	25.99	27.01		
Expeditor	26.10	26.62	27.15		
Expeditor (Fourth Class)	26.64	27.17	27.71		
Operating Assistant (Transferred in house)	24.23	24.71	25.20		
Operating Assistant (Outside hire)	22.02	22.46	22.91		
MAINTENANCE					
Lead Tradesman	30.99	31.61	32.24		
Instrumentation Technologist	33.64	34.31	35.00		
1st Class: Electrician; Instrument Mechanic	30.67	31.28	31.91		
1st Class: Insulator; Mechanic; Machinist/Mechanic; Pipefitter; Welder	30.31	30.92	31.54		
Repairman	25.65	26.16	26.68		
Painter	25.65	26.16	26.68		
Stockkeeper	25.52	26.03	26.55		
Maintenance Office Clerk	25.52	26.03	26.55		
Janitor	23.19	23.65	24.12		
Helper	22.02	22.46	22.91		
Minimum Hiring Rates: (Production, Maintenance, Steam)	21.24	21.66	22.09		

A bonus of 50 cents per hour will be paid to a 3rd or 2nd Class Operating Engineer when acting in the next higher classification and for the position of Assistant Chief Operating Engineer.

Trainer rate is \$ .50/hour above the employee's current rate. This rate is paid only when acting as a trainer.

Relief Shift Supervisor – a bonus of \$ 1.00/per hour above Expediter rate will be paid when acting as a Shift Supervisor.

Maintenance Cell Phone – Trades people requiring to carry a cell phone will be paid \$4.00/hour for the time required.

Students hired to perform jobs in the trades department shall be paid at a rate of not less than 85% of the trades helper rate.

#### Cross-Training Premium

There shall be a  $50\phi/hr$  premium for any employee in the Production Department who is trained and capable to operate one or more additional production processes outside of his home department. For clarity this means an entire process, not a portion thereof.

To qualify the Employee must obtain and continue to maintain competency in the additional process(es) outside of his home department. Competency in a particular process shall be determined by the Process Engineer and evaluated in the normal method, including, among other things, testing and also hands on experience. If an Employee loses competency due to lack of current experience of failure of the testing, or if he is advised that the Company no longer requires the cross training in a particular process, he shall no longer receive the premium.

#### ARTICLE 17 BEREAVEMENT PAY

17.01 In the event of bereavement in an employee's immediate family, the employee will be granted permission to be absent and will be paid for time lost on regular hours during his normal work week up to three (3) consecutive working days up to and including the day after the funeral or memorial service. If the funeral or memorial service is not attended, payment will be made for time lost on only one (1) of the above mentioned days.

Immediate family shall mean current spouse, child, stepchild, mother, father, sister, brother, and the current motherin-law, father-in-law, daughter-in-law, son-in-law, sister-inlaw and brother-in-law.

Permission to be absent will be granted and payment made for one (1) day only for time lost on regular hours during the employee's normal workweek in the event of death of an employee's or the employee's current spouse's, grandchild, grandparent, uncle or aunt or a grandparent of the employee's current spouse.

In the event that interment occurs after the day of the funeral or a memorial service occurs later, permission to be absent and payment made for that day provided that the total allotted time off has not been used.

It will be the responsibility of the employee to make claim for every payment to the Human Resources Department, and to provide evidence satisfactory to the Company to support his claim. Payment for lost time will be at the employee's applicable rate. Regular hours are defined as regular scheduled shifts and overtime shifts. Payment for lost time will be at the employee's applicable rate. Should bereavement occur during a vacation period, the affected vacation days will be rescheduled.

# ARTICLE 18 HURY AND WITNESS SERVICE

18.01 When an employee is required to serve on a jury or as a sub-poenaed witness and is prevented from performing his regular duties because of time required to serve in this capacity, he shall be paid a make-up pay for time lost representing the difference between his fees from the court and his normal earnings for regular hours during the period of such jury service.

#### ARTICLE 19 RENEWAL AND TERMINATION

19.01 This agreement shall become effective June 1, 2015, and remain in full force and effect until May 31, 2018 and shall continue in effect thereafter from year to year for further periods of one year each, unless either party shall have given

written notice of termination or written notice of proposals for amendment to the other not less than sixty (60) days or more than ninety (90) days prior to the expiration of this three year agreement, or any yearly period thereafter.

19.02 In the event of written notice of termination or proposals for amendment having been given by either party as provided in Article 19.01, negotiations shall be carried on during the period of notice with a view of arranging another agreement. Should such negotiations extend beyond the expiration date, this Agreement shall remain in full force and effect until the provisions of the Labour Relations Act of Ontario have been applied or until a new agreement is entered into by the parties

Dated at St Jacobs, Ontario, this 29th day of May, 2015.

Signed on behalf of the parties hereto by their duly authorized representatives:

FOR THE COMPANY: FOR THE UNION R. D. Arndt B. Arnold Manufacturing Manager Negotiation Committee Chair M. Perkovic Hub Manufacturing Manager Negotiation Committee Member Kark Rubio F. Rubie Negotiation Committee Member K. Dickieson Negotiation Committee Member P.D. Lippert Negotiation Committee Member A Bryan

Staff Representative

#### APPENDIX "A" OF COLLECTIVE LABOUR AGREEMENT

#### INTERIM INCREASE

The Cost of Living Allowance (COLA) will be based on the Consumer Price Index (CPI) published by Statistics Canada where 1992 = 100

In each year the COLA will be based on the May CPI. The COLA will trigger after a 5.0 point increase in the CPI All Items has been reached in each year of the collective agreement.

The COLA will be paid on the basis of one cent (\$0.01) for each .15 rise in the CPI All Items after the trigger point has been reached. The COLA will be calculated and paid quarterly and will be rolled into the regular rates as of June 1st of each year of the collective agreement.

Elmira, Ontario

Mr. B. Arnold, Unit President Local Union No. 13691-00, U.S.W. P.O. Box 248 Elmira, Ontario

Dear Mr. Arnold:

This will confirm the commitment made by the Company at the meeting of negotiations on May 29, 2015 concerning payment to employee members of the Bargaining Unit's negotiation committee for time spent in meetings with company representatives for the purpose of negotiation of renewal Agreements between the Company and Union.

Payment will be made at the employee's straight time hourly rate for time spent at every other meeting up to a maximum of 8 hours per day, per committee member, and only up to the date of any application for Conciliation services.

Notwithstanding the above, a Committee member scheduled for a 12 hour shift shall be paid at the employee's straight time hourly rate for time spent at every other meeting up to a maximum of 12 hours per day and only up to the date of any application for Conciliation services.

Sincerely,

Rob Arndt

Manufacturing Manager

Elmira, Ontario

Mr. B. Arnold, Unit President Local Union No. 13691-00, U.S.W. P.O. Box 248

Elmira, Ontario.

Dear Mr. Arnold:

#### Re: Safety Footwear Program

The Company agrees to reimburse employees following the purchase of approved safety boots/shoes on the following basis:

 Effective June 1, 2015 the reimbursement will be based on a maximum of \$ 160.00 for safety boots/shoes in a 12-month period.

Effective June 1, 2016 the reimbursement will be based on a maximum of \$ 165.00 for safety boots/shoes in a 12-month period.

Effective June 1, 2017 the reimbursement will be based on a maximum of \$170.00 for safety boots/shoes in a 12-month period.

The date on the purchase invoice will establish the start of the 12-month period for each individual employee.

- Where an employee requires replacement of worn out safety boots/shoes in a period of less than 12-months from their previous purchase, the maximum reimbursement will be pro-rated by the number of full months elapsed since the previous payment.
- 3. Employees who purchase safety boots/shoes during their probationary period will be eligible for reimbursement only upon completion of the probationary period.

It is understood between the parties that the wearing of approved safety footwear will be mandatory in the plant for all bargaining unit employees.

Sincerely.

Rob Arndt

Manufacturing Manager

Elmira, Ontario

Mr. B. Arnold, Unit President Local Union No. 13691-00, U.S.W. P.O. Box 248 Elmira Ontario

Dear Mr. Arnold:

There have been questions raised regarding the status of employees who, in the vernacular, "blow the whistle" on environmental matters, and we would like to give you the Company's position on the subject.

The Environmental Protection Act provides protection for an employee who reports an environmental problem.

The Company agrees with that principle and will certainly honour the legislation.

The Company also has as its objective the prevention of environmental problems and to that end must rely on employees to report any existing or perceived potential problem to the appropriate member or members of management and again without fear of any repercussions for making such a report.

It is our sincere hope that employees will acknowledge that problems can best be solved within our system.

Sincerely,

Helder Botelho

Hub Manufacturing Manager

#### **Letter of Understanding Payroll Error Correction**

Where an Employee has an error on their pay of \$150 or more, he shall inform the Company not later than 3 business days after the deposit. Upon such notification the Company shall provide a payment to correct the situation within 3 business days of notification.

#### Letter of Understanding Training

The Company and the Union agree that well trained employees are essential for the safe and efficient operation of the Plant and the health and safety of the employees. To meet that end, the Company will select, from employees in the bargaining unit, individual(s) in each department who as part of their duties including training employees. The Company will first solicit volunteers by Seniority before assigning an employee to conduct training.

These employees will work in coordination with other trainers, both bargaining unit and salary employees, within the Company. The applicable training rate will apply when performing these duties, upon prior approval of the Supervisor. Training shall include the training of students in the trades department as set out in 16.03.

#### Letter of Understanding Committees, Teams or Events

The Company will discuss with the Union all new committees, teams, or events that will involve the bargaining unit employees at monthly Union/Management meetings. Committees, teams or events will be added as an agenda item at the monthly Union/Management meeting. The Union may discuss issues with the function of the committees, teams or events and/or involvement of bargaining unit employees at monthly Union/Management meeting. Participation of the employee will be voluntary and will be discussed with the employee beforehand.

It is the Company's intent to involve as many volunteer bargaining unit employees as possible to enhance Company operations and efficiencies. The Company will provide a list of employees involved in such committees, teams or events on an annual basis.

# Letter of Understanding "OHSA Supervisor" Indemnification

The Company recognizes and values the work performed by employees in furtherance of the requirements of the *Occupational Health and Safety Act*, as amended ("OHSA"), and its regulations. The Company and Union agree that certain employees in the bargaining unit may be directed by management to assume some duties and responsibilities that could bring them within the definition of "supervisor", for the purposes of, and as defined in the OHSA and its regulations. This may include, but is not necessarily limited to the positions of Expediter, Lead Tradesman, Lead Shipper, Assistant Chief Operating Engineer, Shift Operating Engineer, Trainer and Relief Shift Supervisor and Senior Lab Technician.

While acting as a supervisor as that term is defined under OHSA, in the event an employee is charged as a supervisor by the Ministry of Labour, or other applicable enforcement agency, with an alleged violation of the OHSA or its regulations, the Company agrees that the employee shall be eligible for Indemnification in accordance with the terms set out herein. To be eligible for Indemnification, the charges must directly relate to acts he performed as a supervisor under the OHSA, and that were within the scope of his job duties and responsibilities, and further provided that he was performing his duties in a diligent, responsible, good faith manner.

Indemnification as set out herein, shall be limited to the following:

- The Company shall provide and pay for legal representation.
- b. The Company shall continue to pay the employee's regular wages during any reasonably necessary time away from work to prepare for and attend trial.
- c. The Company shall, in accordance with its Travel and Expense Policy, pay for reasonably necessary travel and meal expenses to prepare for and attend trial.

#### Letter of Understanding Coverage for Chief Operating Engineer

Coverage for longer vacations (1 week and more)

2<sup>nd</sup> Class Operating Engineers (starting with highest seniority to lowest seniority and sufficient knowledge of the boilers) will be asked whether they are interested in fulfilling the position.

If none of the 2<sup>nd</sup> Class Operating Engineers will cover the following will take place.

3<sup>rd</sup> Class Operating Engineers (starting with highest seniority to lowest and sufficient knowledge of the boilers) will be asked whether they are interested in fulfilling the position.

If none of the 3<sup>rd</sup> Class Operating Engineers will cover the following will take place.

 An employee will be appointed based on their qualifications to perform the duties required to cover for the absence of the Chief Operating Engineer. The least senior qualified Operating Engineer will be assigned to the position.

The requirements of the position are described below:

- The employee will work a minimum of 8 hours per day, up to sufficient hours to equal the same number of hours of pay for that given week.
- The employee will be responsible for the following:
  - Work with the expeditor to ensure maintenance items are addressed
  - Enter the daily steam, water, etc. Information into the computer.
  - Monitor the steam, wastewater, groundwater operations
  - Conduct investigations, non conformances as required
  - Participate in meetings, etc.
  - Review, edit SOP's as requested
  - Update spreadsheets as requested
  - Provide training of employees as directed

#### Coverage for shorter vacations (4 days or less)

 The 2<sup>nd</sup> Class Operating Engineer with the highest seniority to the lowest (from the off shifts and sufficient knowledge of the boilers) will be asked to come in for 8 hours (6:30 to 2:30) If none of the 2<sup>nd</sup> Class Operating Engineers will cover the following will take place.

- The 3<sup>rd</sup> Class Operating Engineer with the highest seniority to the lowest (from the off shifts and sufficient knowledge of the boilers) will be asked to come in for 8 hours (6:30 to 2:30)
- The employee will be responsible for the following:
  - Work with expeditor to ensure maintenance items are addressed
  - Enter the daily steam, water, etc. Information into the computer.
  - Monitor the steam, wastewater, groundwater operations
  - Conduct investigations, non conformances as required
  - Participate in meetings, etc.
  - Work with Environmental Services Supervisor, Remediation Engineering Manager, etc.

If none of the 3<sup>rd</sup> Class Operating Engineers will cover the following will take place.

 The highest class wastewater operator with most seniority will be asked to fulfil the Operating Engineer role. If they decline, the least senior qualified operator will be assigned to the position (the regulation requires as a minimum a 3<sup>rd</sup> Class ticket).

The requirements of the of the position are described below.

- The employee will cover for the Chief Operating Engineer for the eight hours (6:30 to 2:30).
- The employee will be available in the event that a TSSA inspector appears on site. In the event that this person is required for other unanticipated work, he will fulfil that responsibility at the expense of the wastewater/groundwater responsibilities.

It is our understanding that this process does indeed satisfy the requirements ordered by the TSSA in addressing coverage for the Chief Operating Engineer in the event of an absence.

Regarding the rate of pay, based on the level of Operating Engineer, the rate will be equal to the level of the assigned Operating Engineer (second or third class) plus one dollar, multiplied by the standard hours for the week, divided by 40.

#### Letter of Understanding Employee Assistance Program

Chemtura Canada Co./Cie and the United Steelworkers, Local 13691-00 are jointly committed to providing <u>confidential</u> professional assistance to full time employees and their dependants requiring help in dealing with a personal problem.

The matters pertaining to this program will be directed by a joint Company/Union committee of four (4) members; two (2) being selected by the Company and two (2) being selected by the Union.

#### These matters include:

- Maintain and revise, as required, the policies and procedures of the program
- Promote and provide information to employees
- Develop a list of community agencies and resources for employees
- Recommend appropriate assessment services
- Gather statistics on use of the services provide in the program
- Ensure adequate financial resources for the program
- Provide, where required, follow-up via the Health Centre
- Provide training for management, supervisory staff and Union representatives
- Monitor the program as to its effectiveness

The Company will pay for a total of nine (9) counselling sessions per family member per year at a rate of \$ 90.00 per hour or an equivalent thereof in most cases. Upon recommendation by the counsellor, the committee may extend the number of sessions. The cost for any additional sessions beyond this will be the responsibility of the employee. The various counselling agencies invoice the Company directly and do not reveal the identity of the employee or eligible dependent.

Services available include support in the following areas:

- Marriage and Family Relationships
- Alcohol and Other Substance Abuse
- Emotional and/or Psychological Problems
- Financial Guidance, Budgeting and Legal Advice
- Career/Vocational and Pre-Retirement Counselling
- Anger Management and sexual abuse

#### Formal Referral

When a personal or work related problem detrimentally affects an employee's work performance or attendance, appropriate action should be taken to determine the appropriate treatment and continued until its resolution. The decision on the part of the employee to seek help voluntarily is to be commended and will not affect an employee's position or future advancement.

Employees agreeing to participate in the E.A.P. will be expected to cooperate with the recommendations of the treatment professionals to whom they were referred.

Employee's who were formally referred to the E.A.P. will be expected to bring their work performance or attendance back to an acceptable level within a reasonable period of time.

Based on deteriorating job performance, and if there is reason to believe the employee has a personal problem, the company will suggest that the employee make an appointment with a counsellor. If the employee refuses the help that is offered and the job performance or attendance does not improve, or continues to deteriorate, further disciplinary action will be taken.

Counselling can be arranged by contacting any of the agencies listed below:

#### 1. Pace Consulting Group

10 Alpine Court Kitchener, Ontario 519-576-6701 1-800-665-7223 info@pace-eap.com

#### 2. Woolwich Interfaith Counselling Centre

60 Arthur Street South Elmira, Ontario 519-669-8651 info@woolwichcounselling.org

# 3. CARIZON Family and Community Services (MOSAIC previously)

400 Queen St. S. Kitchener, Ontario 519-743-6333 info@carizon.ca

**4.** *K W Counselling Services* 480 Charles St. East Kitchener, Ontario 519-884-0000 1-844-437-3247 info@kwcounselling.com

#### Letter of Understanding Return to Work Program

The Company and the Union fully support the concept of timely and safe reintegration of employees into their regular occupations from both on the job and off the job injuries.

The modified and restructured work or hours of work must be such that it does not risk the aggravation of the injury or illness. The work must be productive and of value to the Company.

Generally, the program will be structured for an employee to be back at regular duties and hours in ten (10) weeks or less. The employee participating in the program is responsible to appraise the Company, on a regular and timely basis, changes in his or her condition by means of the approved Functional Abilities Form (FAF).

The Joint Health and Safety Committee will be kept informed of all employees participating in this program.

Benefits associated with this program include:

- Recovery may be hastened.
- Employees become more productive sooner.
- A skilled employee is back into the work environment.
- Reduced Short Tem Disability and Workplace Safety and Insurance Board (W.S.I.B.) costs.
- Help employees maintain their identity and self-respect.
- Fulfills legal requirements.

#### **Return to Work Committee:**

A joint, equally weighted, Company/Union committee is established to fulfill the purpose and goals of this program. The Committee shall consist of 6 members; three (3) appointed by the Company and three (3) appointed by the Union.

The Committee shall operate under the direction of two (2) cochairs. The Company appointed co-chair or the Union appointed co-chair shall call meetings of the committee as required. It is the responsibility of the committee to determine suitable work assignments for each employee involved in the program. The committee will at all times follow the directions of the health care provider as stipulated on the FAF.

It is recognized that not all members of the committee may be able attend the meetings. However, as a minimum, there shall always be one Company and one Union member present along with the injured or ill employee and his/her supervisor.

#### Responsibilities

#### Company and Union:

The Company and Union will jointly maintain and appraise the program on an on-going basis. The committee will meet as required and discuss the individual needs of an injured or ill employee who qualifies for modified work. Both parties will resolve problems and difficulties that may arise. The Company will provide job descriptions of both the regular and modified jobs to the committee and attending physician.

The Company and Union will inform and promote the program to employees. Confidentiality of personal information will be maintained

#### **Health Care Providers:**

The Company will provide the approved FAF to health care providers. These providers will be actively encouraged to provide clear, accurate and detailed descriptions of the affected employee's capabilities and restrictions.

The right to privacy of the employee's medical information will be maintained.

#### **Employees:**

All employees are eligible to participate in this program. While employees with work related injuries and illness will be required to participate, those suffering from non-work related problems do so on a voluntary basis.

All employees must report work related injuries and illnesses to their supervisor as soon as the employee becomes aware of a problem. Any non-work related injury or illness that may affect work performance must be reported to supervision at the start of the shift.

When seeking medical attention employees must obtain the proper forms, cooperate with any investigations and the completion of all forms, assist with determining appropriate modified duties and ensuring that no other tasks are performed other than those in accordance with the restrictions outlined by the attending physician.

All employees of the Company are encouraged to support and assist fellow employees involved in the return to work program.

Each Case Will Be Reviewed On An Individual Basis.

#### Letter of Understanding Review of WERA AND Group RRSP Language

The Parties discussed in bargaining the need to update the language in the WERA to reflect mandatory language changes under pension law. In addition, the parties discussed the potential of moving the Group RRSP administration to Sun Life. The Bargaining Committee agrees to meet to review the WERA to effect the required changes and to discuss the administration of the Group RRSP, including ensuring compliance with CAP guidelines.

#### Additional Amendments:

Small benefit unlocking rules - adjust to reflect new statutory thresholds

# GROUP LIFE AND HEALTH INSURANCE PLAN

### **BETWEEN**

CHEMTURA CANADA CO/CIE Elmira, Ontario

**AND** 

UNITED STEELWORKERS LOCAL UNION NO. 13691-00 Elmira, Ontario

June 2015 - May 2018

# ALPHABETICAL INDEX GROUP LIFE AND HEALTH INSURANCE PLAN

Subject	Article	Page
Definitions	1	69
Dental Expense Benefit	3	78
Eligibility	2	69
General Provisions	4	80
Group Accidental Death and Dismemberment Benefit	fit3	73
Group Life Insurance Benefit	3	72
Short Term Disability Benefit	3	74
Supplementary Health Benefit	3	76
Supplementary Hospital Expense Benefit	3	76
Termination	5	82

### **PREAMBLE**

The Company agrees to provide the following Group Life and Health Insurance Plan, effective June 1, 2006, under the terms and conditions as detailed in the following Agreement.

# ARTICLE 1 DEFINITIONS

- **1.01** "Employee" means any hourly-paid male or female in regular full-time employment with the Company.
- 1.02 "Dependent" means a person defined as follows:
  - (a) The Spouse of an employee. For purposes of any benefit payable under the Plan, "Spouse" means the person who is legally married to the employee; a common-law spouse who is living with the participant in a conjugal relationship; or former spouse for whom insurance protection for some of the benefits available under the employer's benefit program is mandated by court order.
  - (b) An employee's unmarried, unemployed child under twenty-one years of age.
  - (c) An employee's unmarried, unemployed child who is over 21 years of age but under 25 years of age provided such child is in full time attendance at school, college or university.
- 1.03 "Physician" means a medical practitioner who is registered under the Medical Act of the Province of Ontario or such similar statute or law as governs the practice of medicine in the jurisdiction in which any medical, surgical or diagnostic services are rendered to an employee or his dependant.
- 1.04 "Lay-off" means the termination of employment of an employee by the Company for an indefinite period, caused by the decision of the Company to reduce or eliminate the work upon which the employee was engaged.

# ARTICLE 2 ELIGIBILITY

2.01 All employees actively at work on June 1, 1996, are eligible for benefits covering death, dismemberment or disability suffered or incurred on or after that date (providing the em-

- ployees have fulfilled the initial eligibility requirements as outlined in Section 2.02 following).
- 2.02 New employees are entitled to the benefits of the Group Life and Health Insurance Plan on the first day of the month following the completion of their probationary period..
- 2.03 An employee not actively at work on the dates specified in Sections 2.01 and 2.02 above will be eligible on return to active employment.
- 2.04 (a) An employee will be considered to be single and without dependants until the employee has furnished such proof as the Company may reasonably require to establish the eligibility of any person claimed as a dependant.
  - (b) An eligible dependant of an eligible employee is entitled to benefits as stated herein on or after the date on which the dependant is properly enrolled under the Plan.

If a dependant is confined to hospital on the date benefits would otherwise become effective the coverage will be postponed until the dependant is discharged from hospital.

- (c) Any change in the status of a dependant, which would affect eligibility under the Plan, must be reported promptly to the Company.
- 2.05 Where an insured employee is also insured as a dependant under another group insurance plan for Medical Services, Supplementary Hospital Expense or Supplementary Health, benefits are not payable for the portion of any eligible expense paid by the other plan.

Payment of any eligible expense incurred by an insured dependant, if such dependant is also an insured person under another group insurance plan, will be determined on the following basis:

- (a) Benefits will not be paid for the portion of any expense for which the dependant is eligible to be reimbursed as an insured member of another plan.
- (b) Benefits will be reduced by the amount of reimbursement made by the other plan if the dependant is also insured as a dependant under that plan.

2.06 An employee who is recalled from lay-off or who returns from a properly approved leave of absence, and who retains seniority, shall be eligible for benefit under the plan in respect of any disability incurred by himself or his dependants on or after the date on which he returns to work.

# 2.07 TERMINATION OF ELIGIBILITY

# (a) Voluntary Resignation or Discharge

An employee whose employment is terminated by voluntary resignation or discharge will cease to be eligible for all benefits under the plan on the date of termination, except that life insurance will continue for 31 days under the conversion privilege.

# (b) Lay-off or Retirement

An employee whose employment is terminated by lay-off or by retirement shall cease to be eligible for the Short Term Disability benefit on the date of termination, and for all other benefits on the last day of the month in which termination occurs, except that life insurance will continue for 31 days under the conversion privilege.

An employee whose employment is terminated by lay-off while on an approved leave of absence which was granted for reasons of sickness or injury shall continue to be eligible for Short Term Disability benefit in respect of the disability previously established as the cause of the leave of absence.

# (c) Dependants

Dependants of an employee shall cease to be eligible for benefit under this plan on the date on which the employee ceases to be eligible or on the date on which the dependant no longer qualifies under Section 1.02, and in the case of the death of an employee at the end of the month in which said death occurs.

# 2.08 LEAVE OF ABSENCE - INJURY OR SICKNESS

Subject to the provisions of sub-section 3.03 (g), when an eligible employee is granted leave of absence because of injury or sickness, applicable benefits of the plan for the employee and eligible dependants will be maintained for a per-

iod commencing with the first day of the month following the date such leave commences as follows:

- (a) Up to 6 months for an employee who has less than 5 years' seniority.
- (b) Up to 9 months for an employee who has 5 years but less than 10 years' seniority.
- (c) Up to 12 months for an employee who has 10 or more years' seniority.

### EXPIRATION OF LEAVE OF ABSENCE BENEFITS

An employee who is unable to return to work at the expiration of the period of leave of absence for which he is eligible for benefits as set out above, may continue coverage for himself and dependants for all benefits, except Short Term Disability, by payment, monthly, in advance to the Company, of the total premiums applicable to such benefits. Such payment will be the responsibility of the employee. This privilege will terminate on termination of employment (including lay-off) or termination of the plan or failure to pay premiums.

# 2.09 LEAVE OF ABSENCE – OTHER REASONS

An employee who is granted leave of absence for any reason other than sickness or injury, education or union leave after becoming eligible for the plan, shall continue to be eligible for all benefits for himself and dependants, except Short Term Disability, until the end of the month in which such absence commences.

# ARTICLE 3 BENEFITS

# 3.01 LIFE INSURANCE BENEFIT

Group Life Insurance, with a death benefit payable under the terms and conditions found in a standard Group Life Insurance policy issued in the province of Ontario, will be provided according to the following schedule:

(a) Each employee while eligible for life insurance as an active employee:

Effective June 1, 2006 - \$36,000.00 Effective June 1, 2007 - \$37,000.00 Effective June 1, 2008 - \$38,000.00 (b) Each retiree who becomes eligible for and receives a monthly retirement pension, except a Deferred Vested Pension or a Disability Allowance, from the Company on or after the date specified below:

Effective June 1, 1993 - \$ 7,000.00

If an employee with 10 or more years of service becomes totally disabled prior to age 65 years and is granted a monthly Disability Allowance under provisions of the Wage Employees Retirement Agreement, the life insurance in the amount applicable as an active employee will be extended during the period the employee is in receipt of the monthly Disability Allowance up to age 65 years. At age 65 years the amount of life insurance will be reduced to the amount provided above for a retiree on the date of disability pension.

The group policy will further provide that an individual who has qualified for life insurance as an active employee may, within 31 days of the date of termination of employment, obtain an individual life insurance policy up to a maximum of the amount applicable as an active employee without a medical examination. The group life insurance will continue during this 31 day period.

# 3.02 ACCIDENTAL DEATH AND DISMEMBERMENT BENEFIT

In addition to the group life insurance benefit provided under Section 3.01, a Group Accidental Death and Dismemberment benefit will be provided while eligible as an active employee:

Effective June 1, 2006 - \$36,000.00 Effective June 1, 2007 - \$37,000.00 Effective June 1, 2008 - \$38,000.00

The above principal sum will be payable in the case of death or loss of both hands, both feet, or sight of both eyes, or any two of the above members, providing death or such loss occurs within 90 days of the date of the accident and is caused by external, violent and accidental means.

One-half of the above principal sum will be payable in the case of loss of one hand, one foot, or sight of one eye under the same conditions.

The policy under which this benefit is provided will contain such limitations and conditions as are found in a standard Group Accidental Death and Dismemberment policy issued in the province of Ontario. The maximum payable under this benefit for any one accident is the principal sum.

### 3.03 SHORT TERM DISABILITY BENEFIT

- (a) If, while insured for this benefit, an employee becomes totally disabled so that he cannot work due to a non-occupational accident or illness and is under the care of a physician, payment will be made, on the basis defined in the sub-sections following, up to a maximum of fifty-two (52) weeks during any one period of disability. Appropriate medical documentation from a qualified health care provider is required to be eligible for this benefit. The Company will pay for health care provider notes when required by the Company.
- (b) A period of disability, whether from one or more causes, will be considered as one period of disability and benefits limited accordingly. A recurrence of disability from the same or related causes will be considered a separate period of disability only if the employee has recovered and returned to full time employment for a period of at least fourteen (14) calendar days.
  - A disability resulting from an unrelated cause will be considered a different period of disability if separated by return to regular full time employment.
- (c) If the disability is the result of an accident the benefit will be paid commencing with the first day of disability.

If the disability is the result of illness the benefit will be paid commencing with the fourth (4th) day of continuous disability. If the illness necessitates confinement in a hospital or undergoes day surgery and is unable to return to work, the benefit will be paid commencing with the first (1st) day of hospital confinement if such date is prior to the date on which the benefit would otherwise commence.

The date on which the disability terminates will be deemed to be the day preceding the date on which the employee is capable of returning to work as determined by the attending physician.

(d) Effective June 1, 2001, for claims commencing on or after that date, the weekly benefit is the greater of A and B, subject to a maximum of C, all rounded to the nearest dollar, where:

A = 40 hours times employee's hourly rate times 2/3;

B = average of last 20 weeks of gross pay times 2/3;

C = \$650 (effective June 1, 2015)

It is understood that when there are weeks with less than 16 hours paid, that they will not be included in the average.

Benefit payment for part of a week will be paid at the rate of one-seventh of the Short Term Disability rate multiplied by the number of qualifying days during that week. Total earnings in any week will not exceed the employee's normal pay excluding overtime.

- (e) In consideration of the improvements in the Short Term Disability plan provided in sub-section (d) above, the Union and its members agree to forego the total rebate or savings to which they might be entitled, in whole or in part, under the Employment Insurance Act as a result of the Short Term Disability plan becoming or remaining approved for any reduction of premium, and that all such money shall be retained by the Company.
- (f) Should the provisions of this Section fail to meet or fail to continue to meet the criteria required to maintain the Short Term Disability plan as an approved plan for premium reductions under the Employment Insurance Act, the Company and the Union will, following the effective date of such legislation or change in the regulations, commence negotiations to amend these provisions providing that such amendments do not increase the aggregate cost of Short Term Disability to the Company before the enactment of the legislation or change in the regulations.
- (g) The Short Term Disability benefit is not payable for any disability:
  - (1) During which the employee is not under the care of a physician.

- (2) Resulting from an occupational injury or illness for which the employee receives Workplace Safety and Insurance Board or similar benefit payments.
- (3) Resulting from pregnancy.

However, if the Employment Insurance Commission changes its regulation, generally, to discontinue payment for claims relating to pregnancy, the plan will be revised to provide the payments provided in this Section but shall be limited to a maximum period of 6 weeks.

(4) The Short Term Disability benefit is not payable for any day for which the employee receives payment as a Designated Holiday under the provisions of the Collective Labour Agreement.

# 3.04 SUPPLEMENTARY HOSPITAL EXPENSE BENEFIT

The Company will provide Supplementary Hospital Expense Benefits for any eligible employee and any eligible dependant in the amount of the usual charge of the hospital concerned for standard semi-private ward care decreased by an amount equal to the value of any benefit for which the employee or dependant is eligible under a provincial government hospitalization or health plan. The maximum period for which this benefit will be payable is 365 days in respect of any one Benefit Period.

Successive periods of confinement in a hospital shall be considered as occurring during one Benefit Period unless the employee has recovered and returned to full time employment for a period of at least fourteen (14) days before commencement of the later confinement, or unless the later confinement is due to causes wholly different from those of the prior confinement; or in the case of a dependant, successive periods of confinement are separated by an interval of 3 months or more unless the dependant has completely recovered from the bodily injury or sickness causing the prior confinement before commencement of the later confinement or unless the later confinement is due to causes wholly different from those of the prior confinement.

# 3.05 SUPPLEMENTARY HEALTH BENEFITS

Subject to Eligible Expenses and Limitations, the plan will provide payment of reasonable and customary charges for the benefits listed below for each eligible employee and eligible dependant(s).

Eligible Expenses: Eligible Expenses mean reasonable and customary charges for the following services, providing that such services are prescribed by a physician licensed to practice medicine:

All paramedical services are limited to a combined maximum of \$1,600.00 per year, per patient (\$1,700 effective June 1, 2016). \*Requires a prescription from a physician.

Speech Therapist Psychologist
Chiropractor Osteopath
Podiatrist Naturopath
Acupuncturist Audiologist

Physiotherapist Registered Massage Therapist\*

- Services rendered in the patient's home by a registered nurse, provided such registered nurse is not ordinarily resident in the patient's home and is not a relative of the patient, and subject to maximum total eligible expenses in a calendar year of \$15,000.00 less the amount paid under this provision for registered nurses' services during the two preceding calendar years plus \$50.00 per day thereafter.
- Prescription drugs dispensed by a physician or by a registered pharmacist on the written prescription of a physician. Dispensing fees on prescription drugs will be limited to \$9.00.
- Rental of a wheel chair, hospital bed, iron lung and other durable equipment required for temporary therapeutic use.
- Trusses, braces and crutches where the disability necessitating such equipment is incurred while insured.
- Artificial limbs or other prosthetic appliances where the disability causing loss of the natural member is incurred while insured.
- Local ambulance service.
- The services of a dental surgeon for treatment of a fractured jaw or for the treatment of accidental injuries to natural teeth within six months of the accident where the injury was caused by external, violent and accidental means.

- Services of a physician or surgeon rendered outside Canada, on an emergency basis or following written referral by the attending physician in excess of the charges covered by any government plan.
- Eye glasses up to a total amount of \$275.00 per person in any period of 24 consecutive months when purchased on the written prescription of a medical doctor or optometrist, excluding sun glasses, glasses for cosmetic purposes and safety glasses except where such safety glasses require prescription lenses. Eye exams for employees only will be covered to obtain prescription glasses.
- Hearing aids, including repairs and batteries, up to a total amount of \$500.00 per person in any period of 60 consecutive months, when purchased on the written prescription of a physician.

Limitations: The insurance under this benefit does not cover charges in respect of:

- Injuries or sickness for which benefits are payable under any Workplace Safety and Insurance Act.
- Services of a physician or surgeon except as provided under Eligible Expenses.
- Self-inflicted injuries which are not accidental.
- Injury or sickness resulting from war or from engaging in a riot, illegal disturbance of the peace, or criminal act.
- Any services for which the individual is not required to pay or for which benefits are received under any other group insurance policy.
- Services for which benefits are payable under any government hospital or medical care insurance plan.

### 3.06 DENTAL EXPENSE BENEFIT

The plan will provide for reimbursement of eligible expenses incurred by an eligible employee or eligible dependant as set forth in the following schedule:

- 100% of Basic and Rider No. 1 eligible expenses
- 50% of Rider No. 2 eligible expenses
- 50% of Rider No. 4 eligible expenses

Reimbursement will be on the basis of reasonable and customary charges, and limited to the 2010 Schedule of Fees of the Ontario Dental Association for expenses incurred on or after January 1, 2012; and limited to the 2011 Schedule of Fees of the Ontario Dental Association for expenses incurred on or after January 1, 2013; and limited to the 2012 Schedule of Fees of the Ontario Dental Association for expenses incurred on or after January 1, 2014; and limited to the 2013 Schedule of Fees of the Ontario Dental Association for expenses incurred on or after January 1, 2015.

"Reasonable and customary charges" means the usual charges where there is no insurance.

Employees and dependants will only be covered for dental visits on a 9-month frequency.

# 3.07 EXTENDED TERMINATION BENEFITS

If an employee or an eligible dependant is totally disabled on the date of termination of the employee's insurance because of termination of employment, benefits payable under the Supplementary Hospital Expense Benefit and Supplementary Health Benefit will be continued in respect of the disabled individual during total disability for a period up to 90 days following the date of such termination.

3.08 The following benefits will be provided for a retiree who becomes eligible for and receives a monthly Disability Allowance under provisions of the Wage Employees Retirement Agreement on or after June 1, 1996.

Payment by the Company of the monthly group insurance premiums required to entitle the retiree and any eligible dependants to reimbursement for charges for prescription drugs dispensed by a physician or by a registered pharmacist on the written prescription of a physician.

Payment of such premiums will commence in the month following the effective date of retirement and terminate at the end of the month in which the retiree or spouse reaches age 65 years, or on the death of the retiree, if earlier.

In the case of a Retiree who is receiving a monthly Disability Allowance granted during the term of a previous Life Insurance and Welfare Agreement, the Company will continue to make payment of the monthly premiums required to provide the benefits stipulated in the applicable Agreement.

# ARTICLE 4 GENERAL PROVISIONS

- **4.01** Benefits under this plan are not payable under the following conditions:
  - Injury or illness resulting from self-inflicted injury, act of war or insurrection.
  - (b) Operations or treatments for cosmetic purposes.
  - (c) Operations or treatments for conditions not detrimental to bodily health.
- 4.02 No employee shall be eligible for payment of an indemnity claim either full or in part for any period for which an employee is receiving other income to which the Company is either directly or indirectly a contributor. If an employee, who is temporarily laid off and is receiving Employment insurance, suffers an injury or illness, he would not be excluded from coverage under the plan providing he is otherwise eligible.
- **4.03** No payment of claim will be made if the employee fails to meet the requirements of the insurer with respect to proof and time limitations under regulations included in policies written in Ontario.
- **4.04** An employee suffering an injury or a disability while participating in sport for personal gain and remuneration may be subject to exclusion from any or all coverage under this plan.
- 4.05 The Company may enter into a contract or contracts with an insurance company or companies or such other organizations as it may choose, to provide the benefits of this Agreement, and upon so doing will be relieved of any individual liability to any employee other than to maintain such contract or contracts in force.
- 4.06 If at any time a Federal or Provincial Government passes legislation which directly or indirectly has the effect of providing benefits similar to one or more of the benefits described in this plan for which the employee as a class shall be eligible, this Agreement shall terminate in respect of that

benefit or benefits upon the expiration of thirty (30) days after the proclamation of such statute or upon the date the statute comes into effect, whichever is later.

During such thirty (30) day period or such longer period as may expire after date of proclamation of the statute, the parties will meet for the purpose of negotiating any amendments required or desirable to assure that the aggregate of the statutory benefits and the benefits provided under this plan shall approximate in kind and money value the benefits provided under the plan before said statutory enactment.

# 4.07 CLAIM TO BE FILED WITHIN 90 DAYS

Proof of claim on forms furnished by the Company must be submitted to the Company by the employee within 90 days after the commencement of any disability for which benefits are payable.

4.08 If a dispute shall arise between the Company and the employee, as to whether such employee is, or continues to be suffering from bodily injury or sickness of a degree, extent and type that gives rise to a claim for benefits under the plan, such dispute shall be resolved as follows:

The employee or disabled person shall be examined by a physician appointed for that purpose by the Company and a physician appointed for that purpose by the Union. If they shall disagree concerning the kind and nature of the disability, the question shall be submitted to a third physician, selected by the said two physicians. The opinion of the third physician, after examination of the disabled person and consultation with the other two physicians shall be accepted by the Company, the Union and the employee as irrefutable evidence of the facts herein disclosed, and the degree, extent and type of disability suffered by the disabled person. The fees and expenses of the third physician shall be shared equally by the Company and the Union.

### 4.09 SUBROGATION

When an employee or dependant receives Short Term Disability benefits, or other payments to him or on his behalf by reason of bodily injury or sickness in respect of which some third party is under legal liability, the Company or the insurer at the Company's option, shall be subrogated to the

employee's or dependant's right to compensation for the cost of the benefits and/or services provided in respect of such bodily injury or sickness to the extent of the amount paid by the Company either directly or indirectly, or through coverage provided by the insurance policy in respect thereof and the employee by acceptance of the benefits, will undertake that he or the dependant so entitled to compensation shall prosecute such claim against the third party at the expense of, and to the extent directed by the Company and pay over to the Company what it is entitled to receive as aforesaid together with any expenses it may have paid or incurred, from any monies recovered from such third party, and he or the dependant will do all acts, and execute all documents necessary to permit the Company to obtain the benefit of this clause.

# ARTICLE 5 TERMINATION

- 5.01 This Agreement shall become effective on June 1, 2009, and remain in full force and effect until May 31, 2012, and shall continue in effect thereafter from year to year for further periods of one year each, unless either party shall have given written notice of termination or written notice of proposals for amendment to the other not less than sixty (60) days or more than ninety (90) days prior to the expiration of this three year Agreement, May 31, 2012, or any yearly period thereafter.
- **5.02** In the event of written notice of termination or proposals for amendments having been given by either party as provided for in Section 5.01 of this Article, negotiations shall be carried on during the period of such notice with a view to arranging another agreement.

Should such negotiations extend beyond the expiration date, this Agreement shall remain in full force and effect until the provisions of the Labour Relations Act of Ontario have been applied or until a new agreement is entered into by the parties.

Dated at St. Jacobs, Ontario, this 29th day of May 2015 Signed on behalf of the parties hereto by their duly authorized representatives:

FOR THE COMPANY:	FOR THE UNION
R. D. Arndt	Brias Amold
Manufacturing Manager	Negotiation Committee Chair
Bllefold	Maken Paloco
H. Boteino	
Hub Manufacturing Manager	Negotiation Committee Member
(	Kark Rubie F. Rubie West intin Committee Member
	Negotiation Committee Member
	Affile .
	K. Dickieson
	Negotiation Committee Member
	Maghan
	P.D. Lippert Negotiation Committee Member
	A Bryan
	Staff Representative
	-

# WAGE EMPLOYEES PENSION AND GROUP RRSP AGREEMENT

# **BETWEEN**

CHEMTURA CANADA CO./CIE Elmira, Ontario

AND

UNITED STEELWORKERS LOCAL UNION NO. 13691-00 Elmira, Ontario

June 2015 - May 2018

# ALPHABETICAL INDEX Retirement Plans

Subject	Article	Page
Defined Benefit Plan	1	87
Group RRSP	2	87

### PENSION PLAN

# Article 1 Defined Benefit Plan:

All employees with seniority on June 1, 1996 had their Defined Benefit Plan benefit levels frozen at the basic amount of \$23.00 per month per year of service and the supplemental at \$9.50 per month per year of service. All years of service however will continue. These employees are also enrolled in the Group RRSP.

# Article 2 Group RRSP:

Employees whose Defined Benefit Plan was "frozen" and all new hires are enrolled in the Group RRSP. Contributions to the RRSP plan shall begin following the conclusion of the Employee's probationary period, through regular payroll deduction for the Employee's contributions. The plan is presently managed through AGF Management Ltd. and AIC Group of Funds with the financial advisor being Greg Boehm of Holliswealth. The parties agree that following ratification, they shall discuss the potential transfer of the plan to Sun Life.

Company contributions to the Group RRSP are based on the employee's years of service effective on June 1, 2008.

Years of Service	Percentage
0-12	3.5%
13-20	4.5%
21-25	5.0%
26-30	5.5%
31+	6.2%

Employees will move to the next service grade on their anniversary date

Contributions are based on hours worked. All hours will be at your straight-time hourly rate multiplied by the percentage within the service table. Yearly hours are capped at 2300 hours.

The Company will also contribute when the employee is absent for the following reasons:

- Workplace Safety and Insurance Board
- Short Term Disability
- Pregnancy or Parental Leave
- Bereavement
- Statutory Holidays
- Jury/Witness Duty
- Union Leave of Absence
- Leave of Absence for Apprenticeship Program
- Fire Brigade duties
- Vacation (with payments twice a year)

# **Employee Contributions:**

All employees registered in the Group RRSP are required to contribute to the plan. Contributions will be at 2% of your straight-time hourly rate multiplied by the hours worked. There is no cap on hours worked for employee contributions and you can contribute more than the 2% if you wish.

Dated at St Jacobs, Ontario, this 29th day of May 2015.

Signed on behalf of the parties hereto by their duly authorized representatives:

FOR THE COMPANY: FOR THE UNION R. D. Arndt B. Arnold Manufacturing Manager Negotiation Committee Chair M. Perkovic Hub Manufacturing Manager Negotiation Committee Member Kanh Subis F. Rubie Negotiation Committee Member K. Dickieson Negotiation Committee Member P.D. Lippert Negotiation Committee Member A Bryan

Staff Representative

# WAGE EMPLOYEES RETIREMENT AGREEMENT

# **BETWEEN**

CHEMTURA CANADA CO/CIE Elmira, Ontario

**AND** 

UNITED STEELWORKERS LOCAL UNION NO. 13691-00 Elmira, Ontario

June 2015 - May 2018

# ALPHABETICAL INDEX

# WAGE EMPLOYEES RETIREMENT AGREEMENT

Subject	Article	Page
Administration	15	119
Amount of Retirement Income	7	106
Construction, Interpretation and Definitions	2	94
Contributions and Funding	12	115
Death Benefits	10	111
Disclosure	14	118
Duration and Termination	18	124
Grievance Procedure – Information	16	121
Introduction	1	93
Membership	3	99
Payment of Retirement Benefits	9	109
Protection of Benefits	13	117
Retirement Dates	5	103
Retirement Income Formula	6	105
Revenue Canada Maximum Pension	8	108
Service	4	99
Termination of Employment	11	113
Undertakings	17	122

# ARTICLE 1 INTRODUCTION

- 1.01 This document constitutes the CHEMTURA CANADA Wage Employees Retirement Agreement (the "Retirement Agreement"), being amended and restated herein as of June 1, 2015.
- 1.02 The basic purpose of the Retirement Agreement is to provide retirement income for eligible employees of the Company who were covered as Members under the prior Retirement Agreement immediately prior to June 1, 1996 (hereinafter referred to as the "Prior Retirement Agreement") and who continue to be employed on and after the Effective Date. No person who joins the Company's employment on or after June 1, 1996 shall be permitted to be covered under the terms of the Retirement Agreement. This Retirement Agreement amends and restates the Prior Retirement Agreement with respect to eligible employees and, as such, the document incorporates and preserves the entitlements and benefits accrued prior to the Effective Date under the Prior Retirement Agreement, All assets accumulated under the Prior Retirement Agreement are therefore consolidated in the Fund established for this Retirement Agreement, and all liabilities under the Prior Retirement Agreement are correspondingly assumed under the terms of this Retirement Agreement.
- 1.03 The Retirement Agreement as contained herein shall be applicable to Members who are in the employment of the Company on or after the Effective Date; benefits in respect of a Member whose employment ceased prior to the Effective Date shall be determined in accordance with the terms of the Prior Retirement Agreement at the time of such cessation of employment except as specifically provided herein.
- 1.04 The terms of the Retirement Agreement shall meet the requirements of Applicable Pension Laws and the Revenue Rules and the continued registration of the Retirement Agreement as a registered pension plan under both Applicable Pension Laws and the Income Tax Act is a pre-condition for the Retirement Agreement to become and remain operative.

# ARTICLE 2 CONSTRUCTION, INTERPRETATION AND DEFINITIONS

### CONSTRUCTION AND INTERPRETATION

- 2.01 The masculine pronoun wherever used herein shall include the feminine pronoun where applicable, and the singular shall include the plural and vice versa, as the context shall require. References to a paragraph, Section or an Article, mean a paragraph, Section or an Article in the Retirement Agreement.
- 2.02 The Retirement Agreement and all the rights and obligations hereunder shall be construed, governed and administered in accordance with the laws of the Province of Ontario, except for those rights and obligations which are solely within the jurisdiction of Canada.
- 2.03 All monetary references in the Retirement Agreement are to be construed as being expressed in terms of the lawful currency of Canada.

### DEFINITIONS

In this Retirement Agreement the following terms shall, unless the context clearly indicates otherwise, have the following meanings:

- 2.04 "Actuarial(ly) Equivalent" means a benefit of equivalent value but of different form of payment to a specified benefit, as determined on a basis of calculation adopted by the Company on the advice of the Actuary and in effect on the date such determination is being made, provided that such basis is in accordance with Applicable Pension Laws and provided further that the basis used for purposes of determining the commuted value of a benefit shall be the fair market value of such benefit on the date such determination is being made.
- 2.05 "Actuary" means an individual from time to time appointed by the Company to carry out actuarial valuations and provide such actuarial advice and services as may be required from time to time for the purposes of the Retirement Agreement. The Actuary shall at all times be a person who is a Fellow of the Canadian Institute of Actuaries.
- 2.06 "Applicable Pension Laws" means The Pension Benefits Act of the Province of Ontario and any regulation pursuant thereto and any amendments or substitutes therefore as well

- as any similar statute applicable to the Retirement Agreement and any regulation pursuant thereto adopted by the federal government.
- 2.07 "Basic Benefit Rate" means the Basic Benefit Rate as described in Section 6.01.
- 2.08 "Basic Pension" means the Basic Pension as described in Section 6.01.
- 2.09 "Beneficiary" means that person last designated by the Member to receive any benefit under this Retirement Agreement in the event of the death of the Member according to the provisions of Article 10 or in the absence of an effective designation of a Beneficiary, the estate of the Member.
- 2.10 "Collective Labour Agreement" means the collective labour agreement between the Union and the Company effective June 1, 2006, as amended from time to time.
- 2.11 "Company" means Chemtura Canada Co/Cie. Administrative action required to be taken by the Company shall be performed by the Pension Board Committee in accordance with Article 15.
- 2.12 "Continuous Service" means the service of a Member as defined in Section 4.01, used to determine eligibility for benefits.
- 2.13 "Credited Service" means the service of a Member as defined in Section 4.05, used to determine the amount of benefits for which a Member is eligible.
- 2.14 "Date of Determination" means the date as of which a benefit is to be calculated under the Retirement Agreement, as specified in each relevant Section, and being one of:
  - (a) a Member's Retirement Date,
  - (b) a Member's date of termination of employment,
  - (c) a Member's date of death, and
  - (d) the date of amendment or discontinuance of the Retirement Agreement or the date of consolidation or merger of the Retirement Agreement with another pension plan.
- 2.15 "Disability Retirement Date" means the date specified in Section 5.05.

- 2.16 "Early Retirement Date" means the date specified in Section 5.03.
- 2.17 "Effective Date" means June 1, 2006.
- 2.18 "Employee" means a person who is employed on a regular full-time or regular part-time basis by the Company at the Chemtura Canada Co./Cie, Elmira, Ontario operations and who is a member of the Union or is a member of any other class of employees designated by the Company as being eligible to be covered under the terms of the Retirement Agreement.
- 2.19 "Fund" means the fund established for the purposes of the Retirement Agreement as set forth herein and established in accordance with the terms and provisions of the Funding Agreement, to which all contributions in respect of the Retirement Agreement shall be made and from which all benefits and expenses under the Retirement Agreement shall be payable.
- 2.20 "Funding Agency" means the trust and/or insurance company and/or any group of individual trustees as eligible under Applicable Pension Laws designated by the Company and holding the whole or a portion of the assets of the Fund at any time pursuant to the terms of a Funding Agreement.
- 2.21 "Funding Agreement" means any trust deed, agreement or agreements executed from time to time between the Company and any Funding Agency, including any insurance or annuity contract or contracts issued by a Funding Agency and including any amendments which are from time to time made to any such documents, pertaining to the custody of the investments of the Fund.
- 2.22 "Income Tax Act" means the Income Tax Act (Canada) as amended from time to time, together with any relevant regulations and application rules made thereunder from time to time.
- 2.23 "Member" means an Employee who is covered under the terms of the Retirement Agreement in accordance with Article 3.
- 2.24 "Normal Retirement Date" means the date specified in Article 5.01.

- 2.25 "Old Age Security Date" means the first (1st) day of the month in which the Member first becomes eligible (or could become eligible but for some act or failure to act by him or by another on his behalf, such as leaving Canada, failing to make timely application or failing to notify the proper authority of his address), for benefit under the Old Age Security Act, Canada.
- **2.26** "Pension Board Committee" means the officials of the Company who have been designated and authorized by the Board of Directors of the Company to act on behalf of the Company in accordance with the provisions of Section 15.01.
- 2.27 "Plan-Year" means a twelve (12) month period beginning on January 1 and ending on December 31.
- **2.28** "Postponed Retirement Date" means the date specified in Section 5.04.
- 2.29 "Prior Retirement Agreement" means the agreement established for the Employees, as it existed immediately prior to the Effective Date, and which is hereby succeeded by this Retirement Agreement as of the Effective Date.
- 2.30 "Retirement Agreement" means the Chemtura Canada Wage Retirement Agreement set forth in this document and includes any amendments which are from time to time made hereto.
- **2.31** "Retirement Date" means the Early, Special Early, Normal, Postponed or Disability Retirement Date on which a Member actually retires.
- 2.32 "Revenue Rules" means the provisions of the Income Tax Act pertaining to employees' pension plans or funds registered under the Income Tax Act as they are applicable to this Retirement Agreement.
- **2.33** "Special Early Retirement Date" means the date specified in Section 5.02.
- **2.34** "Spouse" means, at the time a determination is required, a person of opposite sex:
  - (a) who is married to the Member; or
  - (b) who is not married to the Member and is living with such Member in a conjugal relationship for a continuous period not less than three (3) years or in a relationship

of some permanence, if they are the natural or adoptive parents of a child, both as defined in the Family Law Act, 1986 (Ontario) provided that the person is not living separate and apart from the Member at that time.

Where the Date of Determination is on or after December 18, 1999, the foregoing definition of "Spouse" shall be read with the words "of the opposite sex" deleted, subject to the requirements for registration under Applicable Pension Laws and Revenue Rules.

2.35 "Statutory Benefit" means any old age or disability benefit payable under any federal or provincial legislation as now in effect, or under any future federal or provincial legislation, amending, superseding, supplementing, or incorporating existing federal or provincial legislation, but such term does not include Workplace Safety and Insurance Act payments or a benefit payable pursuant to occupational disease laws, the Blind Persons Act, or a benefit payable on a "needs test" basis or solely on account of service in the Armed Forces or other national service, or a benefit payable to or in respect of dependents, or a benefit payable under the Old Age Security Act. Canada.

For the purpose of this Retirement Agreement, the amount of such Statutory Benefit shall be the amount of Statutory Benefit that a Member shall be eligible to receive as of the date he first becomes eligible to receive such Statutory Benefit, and if the Member either does not apply for or loses part or all of such Statutory Benefit through delay in applying for such benefit, by earnings while eligible for such benefit, or other act or failure to act, the amount of the Statutory Benefit may be estimated and the estimated amount applied where the term Statutory Benefit occurs.

In the event that a Member has the option of receiving a Statutory Benefit in a reduced amount on account of receiving it prior to the normal earliest date of eligibility, Statutory Benefit shall mean the amount which would have been payable had such option not been exercised.

- **2.36** "Supplemental Benefit Rate" means the Supplemental Benefit Rate as described in Section 6.02.
- **2.37** "Supplemental Pension" means the Supplemental Pension as described in Section 6.02.

- 2.38 "Union" means United Steelworkers, Local Union No. 13691-00.
- 2.39 "YMPE" means the Year's Maximum Pensionable Earnings established each year under the Canada Pension Plan as amended from time to time or under any superseding legislation considered by the Company to be appropriate.

# ARTICLE 3 MEMBERSHIP

- 3.01 Each person who was a member of the Prior Retirement Agreement immediately prior to the Effective Date shall automatically become covered under the terms of this Retirement Agreement as of the Effective Date and shall be classified as a Member from such date.
- 3.02 No other person, including a person who was an Employee on the Effective Date or who becomes an Employee after the Effective Date, shall become covered under the terms of this Retirement Agreement.
- 3.03 Nothing herein contained shall be deemed to give any Employee the right to be retained in the service of the Company or to interfere with the rights of the Company to discipline, discharge or lay-off any Employee at any time and to treat him without regard to the effect which such treatment might have upon him as a Member.
- 3.04 Upon becoming covered under the terms of this Retirement Agreement, the Employee shall complete and sign the enrolment form provided by the Company, thereby specifying a Spouse and designating a Beneficiary, if necessary.
- 3.05 While a Member remains in employment with the Company, he may not discontinue his active coverage under the terms of this Retirement Agreement.

# ARTICLE 4 SERVICE

# 4.01 CONTINUOUS SERVICE

(a) Continuous Service means the period of uninterrupted, regular, full-time or part-time employment of a Member with the Company, beginning with the date on which he was last hired by the Company and ending on the earliest of:

- his ceasing to be actively employed by the Company where cessation of employment shall include, without loss of generality and subject to Section 4.03, lay-off from employment;
- (2) his death;
- (3) his retirement on pension, including for greater certainty, his retirement on his Disability Retirement Date;
- (4) the discontinuance of the Retirement Agreement without immediate substitution of a successor employees' pension plan.

Continuous Service in respect of an employee whose employment is transferred from temporary to regular, permanent shall include such service as a temporary employee. For greater certainty, such temporary employment shall include periods of vacation employment while pursuing a course of study at a recognized college or university provided that application for such regular, permanent employment is made immediately after completion of such course of study and further provided that only periods worked shall be included in Continuous Service.

- (b) The following shall not constitute interruption of employment, unless and until the Member fails to return to active employment upon expiry of the period concerned:
  - leave of absence duly authorized by the Company, including sickness, accident, emergency, pregnancy or parental leave;
  - (2) total disability qualifying for benefits under Section 5.05;
  - (3) lay-off provided that the Member has not received a benefit pursuant to Article 11 and the period of lay-off does not exceed six (6) months. In the event that such period of lay-off exceeds six (6) months but is less than one (1) year, such lay-off shall not constitute interruption of employment but Continuous Service shall exclude the period of lay-off in excess of six (6) months;

### 4.02 TRANSFER OF EMPLOYMENT

The transfer of a Member within the Company to a category of employment such that the Member ceases to be an Employee for the purposes of this Retirement Agreement, shall not constitute a termination of employment for the purposes of Article 11. In the event of such transfer, the Member's:

- (a) Continuous Service shall include all periods of uninterrupted, regular employment of the Member while he remains in the employment of the Company;
- (b) For the purpose of the computation of the Member's retirement income in Article 6, Credited Service shall exclude those periods of employment while not an Employee. However, for purpose of determining eligibility for benefits under Articles 5 and 11, Credited Service shall include all periods of uninterrupted regular employment of the Member while he remains in the employment of the Company.

# 4.03 RE-EMPLOYMENT

- (a) Subject to paragraph 4.03(b), in the event that an Employee terminates employment other than by retirement, and is subsequently re-employed with the Company, his periods of Continuous Service shall be treated separately, and the second period shall be considered to start from the date of said subsequent re-employment for the purposes of the Retirement Agreement.
- Different periods of prior Continuous Service of five (b) (5) years or more, together with prior Continuous Service for any intervening period or periods of service of less than five (5) years which were terminated otherwise than by voluntary resignation or discharge shall be combined with Continuous Service after re-employment and treated as one continuous period of Continuous Service for an Employee who, at his date of termination of employment other than by retirement, had five (5) or more years of Continuous Service to his credit and who subsequent to his date of re-employment remains employed with the Company for a continuous period of at least five (5) years. For greater certainty, the period during which the Employee was not employed by the Company shall be excluded in the cal-

culation of Continuous Service. This paragraph 4.03(b) shall not apply to an Employee who has received a benefit pursuant to the terms of Article 11.

4.04 In the calculation of Continuous Service, a period of fifteen (15) days or more shall constitute one (1) month of Continuous Service; a period of fourteen (14) days or less shall be disregarded in its entirety.

### 4.05 CREDITED SERVICE

- (a) Subject to Section 4.06, Credited Service with respect to a Member means the Member's Continuous Service.
- (b) Notwithstanding any other provisions of the Retirement Agreement, in respect of eligible periods of employment during which the Member was employed regularly on a part-time basis, Credited Service shall be reduced to an amount which will have the same proportionate relationship as the regular hours of such a Member have to the normal hours of a Member in full-time employment.

### 4.06 EXCLUSIONS FROM CREDITED SERVICE

Notwithstanding the provisions of Section 4.05, Credited Service shall not include:

- (a) any period of active membership in any other plan of the Company for which a benefit is credited under such other plan, except as provided in the Retirement Agreement:
- (b) any period of temporary absence as described in subparagraph 4.01(b)(1) during which such Member is not receiving earnings from the Company or income under Section 5.05 unless the Member returns to employment with the Company prior to Normal Retirement Date upon cessation of disability. This paragraph does not apply to short periods of leave of absence not exceeding one (1) year, unless the Member fails to return to active employment, in which case such leave of absence will not be included in Credited Service.

In no event, however, shall the total periods of temporary absence to be included as Credited Service, excluding those throughout which the Member suffers a physical or mental impairment, as certified in writ-

ing by a qualified medical doctor licensed to practice under the laws of Canada, that prevents the Member from performing the duties of employment in which the Member was engaged before the commencement of the impairment, exceed the sum of:

- (1) five years; and
- (2) the periods of parenting, as defined in Revenue Rules, subject to a maximum of 36 months of such periods of parenting and a maximum of 12 months for any one period of parenting.

# ARTICLE 5 RETIREMENT DATES

# **5.01** NORMAL RETIREMENT

The Normal Retirement Date of a Member is the first (1st) day of the month next following the attainment of age sixty-five (65).

# 5.02 SPECIAL EARLY RETIREMENT

A Member may retire or may be retired on a Special Early Retirement Date which shall be the first (1st) day of any month following or coincident with the Member having both attained age sixty-two (62) and completed ten (10) years of Credited Service.

# 5.03 EARLY RETIREMENT

A Member may retire or may be retired on an Early Retirement Date which shall be the first (1st) day of any month following or coincident with the Member:

- (a) having attained age fifty-five (55); or
- (b) having attained age fifty-five (55), but not age sixty-two (62), and having combined years of age and Credited Service total eighty-five (85); or
- (c) having attained age sixty (60), but not age sixty-two (62), and having completed ten (10) years of Credited Service.

# **5.04** POSTPONED RETIREMENT

If a Member continues in the employ of the Company beyond Normal Retirement Date, the Member shall retire, or be deemed to have retired for the purposes of the Retirement Agreement, not later than the first (1st) day of the month immediately prior to the Member's attainment of age sixty-nine (69). The date of the Member's actual retirement in accordance with this paragraph shall be his Postponed Retirement Date.

#### 5.05 DISABILITY RETIREMENT

- A Member, who through no fault of his own as described below, suffers a total and permanent disability, shall be retired, provided he has completed ten (10) years of Credited Service. For this purpose, a Member shall be deemed by the Company to be totally and permanently disabled if it is certified by a qualified medical doctor who is licensed to practice under the laws of Canada that, as a result of bodily or mental injury or disease, he is wholly and presumably permanently prevented from engaging in any regular occupation or employment for remuneration, except that no Member shall be deemed to be totally and permanently disabled if such disability resulted from engaging in a criminal enterprise, or an intentionally self-inflicted injury, or injuries received due to service in the armed forces of any country, or if such disability resulted from or consists of chronic alcoholism or addiction to narcotics. The date from which such certification is effective shall be deemed to be the Member's Disability Retirement Date.
- (b) A Member in receipt of retirement income as a result of disability is required from time to time, but not more frequently than at six (6) month intervals, to report at a reasonable place for physical re-examination. In the event that a Member refuses or neglects to be so re-examined, the Member's disability retirement income shall be discontinued until such time as the continuance of his disability, as described in paragraph 5.05(a) is so verified by such examination.
- (c) In the event that a Member in receipt of retirement income ceases to be totally and permanently disabled as described in paragraph 5.05(a) or engages in any work similar to that described in the job requirements of any job in the bargaining unit as defined in the Collective Labour Agreement, the retirement income in respect of such a Member may be terminated at the Company's

discretion and such Member shall be rehired by the Company in a capacity consistent with his seniority and his physical and mental ability, provided the Member promptly applies for such work.

## ARTICLE 6 RETIREMENT INCOME FORMULA

#### 6.01 BASIC PENSION

The monthly amount of Basic Pension as of a Member's Date of Determination shall be equal to the Basic Benefit Rate multiplied by his years of Credited Service.

For all Members, the Basic Benefit Rate shall be \$23.00.

### 6.02 SUPPLEMENTAL PENSION

(a) The monthly amount of Supplemental Pension as of a Member's Date of Determination shall be equal to the Supplemental Benefit Rate, multiplied by the Member's years of Credited Service to a maximum of thirty (30) such years. The Supplemental Benefit Rate shall depend on the Member's Date of Determination and shall be as follows:

Date of Supplemental
Determination Benefit Rate
On or after June 1 1996 \$9.50

The Supplemental Pension shall be reduced by the amount of Statutory Benefit to which the Member may be or becomes eligible for prior to his Old Age Security Date.

Notwithstanding the foregoing, the Supplemental Pension shall not be reduced by the amount of Statutory Benefit which the Member may be eligible to or become eligible to receive in a reduced amount, at his option, prior to the normal earliest date of eligibility for such Statutory Benefit.

(b) The Supplemental Pension shall be payable in the form of monthly instalments beginning on the last day of the month in which the Member's Retirement Date occurs and continuing thereafter up to and including the month in which the Member's Old Age Security Date occurs or his earlier death.

## ARTICLE 7 AMOUNT OF RETIREMENT INCOME

#### 7.01 NORMAL RETIREMENT

A Member who retires on his Normal Retirement Date will receive an amount of retirement income computed in accordance with Section 6.01, using the Member's Normal Retirement Date as his Date of Determination.

### 7.02 SPECIAL EARLY RETIREMENT

- (a) A Member who retires on a Special Early Retirement Date will receive an amount of retirement income computed in accordance with Section 6.01 using the Member's Special Early Retirement Date as his Date of Determination.
- (b) In addition to the retirement income in paragraph 7.02(a), a Member who retires on a Special Early Retirement Date will receive a Supplemental Pension computed in accordance with Section 6.02 using the Member's Special Early Retirement Date as his Date of Determination.

#### 7.03 EARLY RETIREMENT

- (a) A Member who retires on an Early Retirement Date under paragraph 5.03(a) will receive an amount of retirement income computed in accordance with Section 6.01 using the Member's Early Retirement Date as his Date of Determination, reduced to the lesser of:
  - the Actuarial Equivalent of the retirement income otherwise payable at his Normal Retirement Date;
     and
  - (2) the retirement income otherwise payable at his Normal Retirement Date, reduced by 0.25% for each month by which the pension commencement date precedes his Normal Retirement Date.
- (b) A Member who retires on an Early Retirement Date under either paragraph 5.03(b) or paragraph 5.03(c) will receive an amount of retirement income computed in accordance with Section 6.01 using the Member's Early Retirement Date as his Date of Determination,

multiplied by the applicable early retirement factor, in accordance with paragraph 7.03(d).

- (c) In addition to the benefit in paragraph 7.03(b), a Member who retires on a Early Retirement Date under either paragraph 5.03(b) or paragraph 5.03(c) will receive a Supplemental Pension computed in accordance with Section 6.02 using the Member's Early Retirement Date as his Date of Determination, multiplied by the applicable early retirement factor, in accordance with paragraph 7.03(d).
- (d) The early retirement factor referred to in paragraphs 7.03(b) and 7.03(c) shall be determined by the following table:

Period in Months by Which	
Pension Commencement	Percentage
Precedes Age Sixty-Two (62)	Payable
0	100.0
12	95.2
24	90.4
36	85.6
48	80.8
60	76.0
72	71.2
84	66.4

For periods not shown above, the appropriate percentage payable shall be determined on a pro-rata basis.

### 7.04 POSTPONED RETIREMENT

A Member retiring on a Postponed Retirement Date shall be entitled to receive an amount of retirement income payable from the Member's Postponed Retirement Date and computed in accordance with Section 6.01 using the Member's Postponed Retirement Date as his Date of Determination.

#### 7.05 DISABILITY RETIREMENT

(a) A Member who retires on his Disability Retirement Date will receive an amount of retirement income computed in accordance with Section 6.01, using the Member's Disability Retirement Date as his Date of Determination and, subject to paragraph 7.05(d), commencing on his Disability Retirement Date.

- (b) In addition to the retirement income in paragraph 7.05(a), a Member who retires on his Disability Retirement Date will receive a Supplemental Pension computed in accordance with Section 6.02 using the Member's Disability Retirement Date as his Date of Determination and, subject to paragraph 7.05(d), commencing on his Disability Retirement Date.
- (c) A Member who is in receipt of retirement income under Section 7.05 of the Retirement Agreement shall not be entitled to any other retirement income or allowance under the Retirement Agreement.
- (d) Notwithstanding the foregoing, retirement income in the event of disability shall not commence during a period when the Member is on approved leave of absence from the Company and is in receipt of or is eligible to receive indemnity benefits from an insurance or compensation plan to which the Company contributes.

# ARTICLE 8 REVENUE CANADA MAXIMUM PENSION

- 8.01 Anything to the contrary contained herein notwithstanding, the maximum amount of lifetime annual pension provided to any employee or former employee under this Retirement Agreement and any other pension plan of the Company shall not exceed the product of (a) and (b) where:
  - (a) is the lesser of:
    - two percent (2%) of the average of the best three
       years of remuneration of the employee; and
    - (2) \$1,722.22 or such higher amount as permitted under the Income Tax Act and the regulations thereunder,

and

- (b) is the sum of:
  - (1) the lesser of
    - (a) pensionable service prior to January 1, 1992;and
    - (b) thirty-five (35) years, and
  - (2) pensionable service after December 31, 1991; where pensionable service is the service permitted under the Income Tax Act and the regulations thereunder.

## ARTICLE 9 PAYMENT OF RETIREMENT BENEFITS

#### 9.01 NORMAL FORM

Unless a Member elects an optional form of Basic Pension under Section 9.03, payment thereof shall be in the form of monthly instalments beginning on the last day of the month in which his Retirement Date occurs and continuing thereafter up to and including the month of death, or earlier cessation of disability (as defined in paragraph 5.05(c)).

Notwithstanding the foregoing, the Basic Pension payable to a Member who retires on his Normal Retirement Date and retirement income payable under Section 11.02 which is locked-in pension as defined under Applicable Pension Laws, will be payable as above, with a guarantee that not less than sixty (60) monthly payments shall be made to the Member and the Member's Beneficiary combined.

## 9.02 STATUTORY SPOUSAL PENSION

Notwithstanding Section 9.01, in respect of Basic Pension. a Member who has a Spouse at the Member's Date of Determination and who is entitled to retirement income under the Retirement Agreement shall elect, or shall for purposes of the Retirement Agreement be deemed to have elected, an optional form of retirement income under which the Member shall receive the Actuarial Equivalent of the retirement income described in Section 9.01 payable during the Member's lifetime with the provision that, on his death, sixty percent (60%) of such actuarially reduced retirement income shall be continued to his surviving Spouse during the Spouse's lifetime. A Member may elect to receive his Basic Pension in the normal form as described in Section 9.01 provided a waiver has been signed in a manner prescribed under Applicable Pension Legislation. A Member may elect an alternative form of pension payment under Section 9.03 provided such waiver has been signed.

### 9.03 ELECTION OF OPTIONAL GUARANTEED FORM

(a) With the approval of the Company, a Member who is retiring at his Normal Retirement Date and who is entitled to a guaranteed retirement income under Section 9.01 may elect, in lieu of such retirement income payable in the normal form as described in Section 9.01, any optional form of retirement income contained in this Section 9.03 as may be applicable. Such written election in prescribed form must be filed with the Company prior to his attainment of age sixty (60). Such written election may be filed with the Company after the Member's attainment of age sixty (60) and prior to his Normal Retirement Date provided that the Member accompanies such election with evidence of good health which is satisfactory to the Company. The retirement income payable under the elected option shall be the Actuarial Equivalent of the retirement income payable under the normal form as described in Section 9.01.

- (b) An election made under paragraph 9.03(a) may be revoked by the Member at any time prior to the Member's Normal Retirement Date. Such revocation shall be made by filing written notice thereof with the Company. Subsequent elections under paragraph 9.03(a) shall not be permitted after such revocation.
- (c) In the event of the death of the Member prior to his Normal Retirement Date, elections under paragraph 9.03(a) shall be considered to be inoperative and ineffective.

## (d) Life Annuity – Guaranteed Term

A Member may elect to receive retirement income payable to him during his lifetime, with the provision that should he die before he has received payments for the available guaranteed term (in complete years only) elected by him, then either the remainder of the payments shall be paid to his Beneficiary or to the estate. At the election of the Member's estate or, if applicable, the Beneficiary's estate, the Actuarial Equivalent Value of the remaining payments shall be paid to the applicable estate. The guaranteed terms available are one hundred and twenty (120) months and one hundred and eighty (180) months.

#### 9.04 SMALL BENEFITS

Effective July 1, 2012, if the annual retirement income payments at normal retirement date under the Retirement Agreement are less than four percent (4%) of the YMPE as

at the Date of Determination, or if the Actuarial Equivalent of such retirement income payments does not exceed twenty percent (20%) of the YMPE as at the Date of Determination, the Pension Board Committee may direct, subject to Section 9.06, the payment of the Actuarial Equivalent in a lump sum to the recipient.

## 9.05 COMMUTATION FOR SHORTENED LIFE EXPECTANCY

Retirement income currently being paid or required to be paid under the Retirement Agreement may be commuted and paid in a lump sum at the discretion of the Member if the Member:

- (a) establishes that he has an illness or physical disability that is likely to shorten his life expectancy to less than two (2) years, as certified by a written statement from a qualified medical doctor licensed to practice in Canada:
- (b) provides an application to the Company in the prescribed form; and
- (c) satisfies any other conditions prescribed by Applicable Pension Laws.

### 9.06 PAYMENTS TO RRSP

Effective July 1, 2012, where a Member or surviving Spouse has an entitlement under this Retirement Agreement to a lump sum payment that is not locked-in, that person may direct the Pension Board Committee, in accordance with and within the time prescribed by Applicable Pension Laws, that such entitlement be transferred to a registered retirement savings plan or a registered retirement income fund of which he or she is the annuitant, provided the transfer complies with Applicable Pension Law and the Revenue Rules.

## ARTICLE 10 DEATH BENEFITS

## 10.01 DEATH BENEFITS PRIOR TO NORMAL RETIREMENT DATE

If the death of a Member occurs prior to the earlier of the Member's Retirement Date and Normal Retirement Date the following benefit is payable to his Spouse, unless waived in prescribed form in accordance with the Applicable Pension Laws:

The Actuarial Equivalent of the Basic Pension that the Member would have been eligible to receive in accordance with Section 7.01, Section 7.02, Section 7.03 or that the Member would have been eligible to receive in accordance with Section 11.02, as applicable,

- in respect of Credited Service on and after January 1, 1987, and,
- (2) in respect of Credited Service prior to January 1, 1987 where such Basic Pension is the improvement resulting from amendments made to the Retirement Agreement on and after January 1, 1987, assuming that the Member retired or terminated employment, as applicable, on the date of death.

At the Spouse's election, the benefit shall be payable as an annuity commencing on or before the Spouse's attainment of age sixty-five (65) or as a lump sum. If the Member has no Spouse at date of death, the above benefit shall be payable in a lump sum to his Beneficiary.

## 10.02 DEATH BENEFITS AFTER NORMAL RETIREMENT DATE AND BEFORE POSTPONED RETIREMENT

- (a) If the death of a Member occurs after the Member's Normal Retirement Date, but before the Member's Postponed Retirement Date, for the purpose of determining any death benefit payable hereunder, the Member shall be deemed to have retired on the first (1st) day of the month coincident with or immediately preceding the date of the Member's death.
- (b) Nothwithstanding paragraph 10.02(a), a Member's Spouse or Beneficiary shall be entitled to receive such larger amount, if any, as specified in Section 10.01 with the references to Section 7.01, Section 7.02, Section 7.03, and Section 11.02 replaced by Section 7.04.

## 10.03 DEATH BENEFITS AFTER RETIREMENT DATE

If the death of a Member occurs after his Retirement Date there shall be paid to his Beneficiary any benefits due in accordance with the retirement income option elected by the Member under Article 9. In the case of a Member who did not elect an optional form of retirement income, payment shall be made to the Member's Beneficiary in accordance with Section 9.01 or Section 9.02 as applicable.

#### 10.04 BENEFICIARY DESIGNATION

A Member shall designate in writing a Beneficiary to receive any benefits payable under the Retirement Agreement upon the death of such Member and may change such designation from time to time. Such designation or change must be in accordance with any law applicable to the Member and shall be in such form and executed in such manner as the Company may, from time to time, determine. Any designation or change must be filed with the Company. In the absence of an effective designation of a Beneficiary, the Company shall instruct the Funding Agency to make payment of any death benefits under this Retirement Agreement to the estate of the Member and any such payment shall completely discharge all liability with respect to the amount paid. Any Beneficiary designations under the Prior Retirement Agreement that are on record with the Company as of the Effective Date shall remain in effect for the purposes of this Retirement Agreement.

## ARTICLE 11 TERMINATION OF EMPLOYMENT

- 11.01 A Member whose employment with the Company is terminated prior to July 1, 2012 and prior to the completion of two (2) years of Credited Service, for any reason other than death or retirement, shall not be eligible to receive any benefit under the terms of this Retirement Agreement.
- 11.02 (a) A Member whose employment with the company is terminated on or July 1, 2012 for any reason other than death or retirement (including disability retirement as described in Section 5.05) is entitled to receive retirement income equal to the basic benefit rate applicable to the Member as at January 1, 1987 under the terms of the pension agreement at such time multiplied by the Member's years of Credited Service prior to the first (1st) day of January, 1987.
  - (b) In addition to the amount in paragraph 11.02(a), a Member whose employment with the Company is terminated on or after July 1, 2012 for any reason other than death or retirement (including disability retirement as described in Section 5.05), is entitled to receive retirement income:

- (1) in respect of Credited Service on and after January 1, 1987, and
- (2) in respect of Credited Service prior to January 1, 1987 where such retirement income is the improvement in the Basic Pension resulting from amendments made to the Plan on and after January 1, 1987, determined in accordance with Section 6.01 using the Member's date of termination of employment as the Date of Determination.
- The retirement income under paragraph 11.02(a) and (c) paragraph 11.02(b) is payable in accordance with Article 9.01 and Article 9.02, commencing at the Members' Normal Retirement Date. At the Member's election. the retirement income may commence on the first (1st) day of any month following attainment of age fifty-five (55). In this case, the retirement income will be reduced to the Actuarial Equivalent of the retirement income otherwise payable at Normal Retirement Date. In no event however, shall such reduction be less than the reduction required by paragraph 8503(3)(c) of the Income Tax Regulation. Application for such retirement income must be made in writing by the Member not earlier than sixty (60) days prior to his elected date of commencement of such income
- 11.03 A Member entitled to receive retirement income in accordance with Section 11.02 or the Spouse of a deceased Member who is entitled to retirement income under the terms of the Retirement Agreement may direct that the lump sum that is the Actuarial Equivalent of such pension:
  - (a) be transferred to another employee's pension plan; or
  - (b) be transferred to a registered retirement savings plan or such other registered vehicle as may be approved under Applicable Pension Laws and Revenue Rules, in the name of the Member, or Spouse where applicable, or
  - (c) be applied towards the purchase of an immediate or deferred life annuity from an insurance company; provided, however, that the administrator of such plan or vehicle agrees in writing to administer such transferred benefit as a deferred life annuity within the conditions of Applicable Pension Laws.

The transfers under this Section 11.03 shall be subject to any limitations prescribed by Applicable Pension Laws and Revenue Rules in respect of the transfer of monies from the Fund

11.04 Effective July 1, 2012, notwithstanding any other provision of the Retirement Agreement, in the event that a Member is entitled to grow-in benefits as provided by Applicable Pension Laws his or her pension will be determined as prescribed by Applicable Pension Laws."

## ARTICLE 12 CONTRIBUTIONS AND FUNDING

#### **12.01** FUND

- (a) The retirement income and other benefits provided under the terms of the Retirement Agreement, shall be financed by a Fund established for the purposes of the Retirement Agreement under which all contributions and investment income are held to pay such retirement income, other benefits and expenses.
- The Company shall be responsible for the selection of (b) a Funding Agency, and its appointment by the Board of Directors of the Company. The Fund or a portion thereof shall be maintained and administered by the Funding Agency in accordance with the terms of a Funding Agreement entered into between the Company and the Funding Agency. The Company and the Funding Agency may agree to amend the form and the terms of the Funding Agreement at any time. The Board of Directors of the Company may further appoint an organisation licensed to provide investment management services, to manage the investment of any portion of the Fund. The Board of Directors of the Company may replace any Funding Agency or investment manager at any time, in accordance with the terms of any applicable agreement or contract.
- (c) Subject to Applicable Pension Laws, the retirement income and other benefits provided under this Retirement Agreement payable hereunder shall only be paid to the extent that they are provided for by the assets held under the Fund, and no liability or obligation to make any contributions thereto or otherwise shall be

imposed upon the Funding Agency or the Company other than in accordance with Section 12.02.

- (d) The investment of the Fund shall be made in accordance with Applicable Pension Laws and Revenue Rules so as not to subject the Fund to income tax liability.
- (e) Fees of the Funding Agency, fees of an investment manager, investment brokerage, transfer taxes and similar costs arising as a result of the making of investments, sale of assets or realization of investment yield, and the expenses reasonably incurred or compensation properly paid in the course of the administration of the Retirement Agreement, including actuarial and pension consulting fees, shall be paid by the Company in cash or from the Fund as determined by the Company.

### 12.02 COMPANY CONTRIBUTIONS

Based upon the amounts estimated by the Actuary, the Company will contribute to the Fund such amounts as are necessary after application of actuarial surplus determined by actuarial valuation, or any portion thereof, to provide for the benefits accruing in that year and to fund any unfunded liability and any experience deficiency in accordance with, and within the time limits specified in, Applicable Pension Laws. Subject to Applicable Pension Laws, the liability of the Company at any time is limited to such contributions as should have theretofore been made by it in accordance with Applicable Pension Laws. The Company contributions shall be in accordance with Applicable Pension Laws.

Notwithstanding the foregoing, contributions made to the Fund after 1991 by the Company shall only be made if they are eligible contributions in accordance with the Income Tax Act and the rules and regulations thereunder. In the event that the Company makes a contribution to the Fund, which would cause the revocation of the Retirement Agreement's registration under the Income Tax Act then, subject to conditions or approval procedures under the Pension Benefits Act of Ontario, such contributions shall be returned to the Company.

#### 12.03 CLAIMS ON THE FUND

No Member or any person claiming through him, by virtue of any provision of this Retirement Agreement, shall have any right to, or any interest in, any part of the Fund except to the extent provided from time to time under the Retirement Agreement and the Funding Agreement, and any Member or other person having any claim through him shall have recourse solely to the Fund for payment of any benefits hereunder. Under no circumstances shall any liability attach to the Company or Funding Agency, or any director, officer or employee of the Company for payment of any benefits or claims hereunder.

## ARTICLE 13 PROTECTION OF BENEFITS

- 13.01 Except as permitted under Section 13.02 and subject to Applicable Pension Laws and the portability and commutation provisions of any other Article of the Retirement Agreement, no benefit, right or interest provided under the plan shall be capable of anticipation, alienation, sale, transfer, assignment, surrender, pledge, encumbrance, charge, seizure, attachment, being given as security or other legal or equitable process. For the purpose of this Section 13.01, "surrender" does not include a reduction in benefits to avoid the revocation of the registration of the Retirement Agreement under the Income Tax Act.
- 13.02 (a) Subject to Applicable Pension Laws and pursuant to a written agreement, decree, order or judgement of a competent tribunal, a benefit payable under the Retirement Agreement may be subject to execution, seizure or attachment in satisfaction of an order for support or maintenance or may be assigned, pledged, charged, encumbered or alienated to satisfy a division of matrimonial property on or after the breakdown of the marriage in settlement of rights arising out of the marriage.
  - (b) The determination of the benefit payable to a person under paragraph (a) shall be subject to Applicable Pension Laws and Revenue Rules.
  - (c) The Member's benefit entitlements shall be reduced to account for the value of any settlement made under paragraph (a). Such reduction shall be determined in accordance with Applicable Pension Laws and Revenue Rules.

- **13.03** If the Company shall receive evidence which in its absolute discretion is satisfactory to it that:
  - (a) a person entitled to receive any payment provided for under the terms of the Retirement Agreement is physically or mentally incompetent to receive such payment and to give a valid release therefore,
  - (b) another person or an institution is then maintaining or has custody of such payee, and
  - (c) no guardian, committee or other legal representative of the estate of such payee shall have been duly appointed, then the Company may direct the payment to such other person or institution, and such payment shall be a valid and complete discharge to the Retirement Agreement for the payment.

In the absence of the appointment of a legal guardian, any benefit payable to a minor may be paid to such adult or adults as have, in the absolute discretion of the Company, assumed the custody and principal financial support of such minor.

## ARTICLE 14 DISCLOSURE

## 14.01 RETIREMENT AGREEMENT COPY

On or before the date an Employee becomes eligible for coverage under the terms of the Retirement Agreement, the Company shall provide to such Employee a copy of the Retirement Agreement.

Not later than allowed under Applicable Pension laws, the Company shall provide a copy of the amendment or the revised agreement to each Employee affected by the amendment.

### 14.02 INSPECTION

The Company shall permit a Member or such other person as is required to be permitted by Applicable Pension Laws (including an authorized agent) to inspect, or make extracts from, documents required to be made available under Applicable Pension Laws, at any reasonable time at such offices of the Company including its principal office as designated by the Company.

### 14.03 BENEFITS STATEMENT

- (a) Within six (6) months after the end of each Plan Year, the Company shall provide to each Member a written statement describing the benefits the Member has earned to date and such other information as required under Applicable Pension Laws.
- (b) Upon cessation of employment of a Member or upon termination of his active coverage under the terms of this Retirement Agreement, the Company shall provide the Member (or the person entitled to benefit in the event of the Member's death) with a written statement of the benefits to which he is entitled.

#### 14.04 OTHER INFORMATION

The Company shall provide such other information regarding the Retirement Agreement, statistical or otherwise, as is required under Applicable Pension Laws.

### 14.05 LIMITATION

Such statement or other information provided shall have no effect on the rights or obligations of any person under the Retirement Agreement, and shall not be referred to in interpreting or giving effect to the provisions of the Retirement Agreement. The Company shall not be liable for any loss or damage claimed by any person to have been caused by any error or omission in such statement or other information.

## ARTICLE 15 ADMINISTRATION

- 15.01 The Retirement Agreement shall be administered by the Company. To facilitate any action required to be taken by the Company under the provisions of the Retirement Agreement, the Board of Directors of the Company has established the Pension Board Committee to act on behalf of the Company.
- 15.02 The Pension Board Committee may from time to time direct that appropriate records be maintained and may establish rules for the administration of the Retirement Agreement. The Pension Board Committee shall have the exclusive right to interpret the Retirement Agreement provisions and to decide any matters arising hereunder in the administration and

operation of the Retirement Agreement. All interpretations and decisions shall be applied as nearly as may be possible in a uniform manner to all Members similarly situated.

- **15.03** The Pension Board Committee is empowered to execute certificates of amendments to the Retirement Agreement.
- 15.04 The Company shall indemnify and save harmless the members of the Pension Board Committee and any other employees who are involved in the administration of the Retirement Agreement from the effects and consequences of their acts, omissions and conduct in their formal capacity to the extent permitted by law except for their own wilful and intentional malfeasance or misconduct. No part of the Fund shall be used for indemnification payments.
- 15.05 The Company and members of the Pension Board Committee shall be entitled to rely conclusively upon all tables, valuations, certifications, opinions and reports which shall be furnished by an actuary, accountant, legal counsel or other professional person who shall be employed or engaged for such purposes.
- **15.06** Whenever the records of the Company are used for the purposes of the Retirement Agreement, such records shall be conclusive of the facts with which they are concerned.
- 15.07 An eligible Employee, a Member, a Beneficiary, or a Spouse shall sign such application forms prescribed by the Pension Board Committee and furnish proof of age and furnish such other data and sign such documents as the Pension Board Committee deems necessary or desirable for the proper administration of the Retirement Agreement or to evidence initial or continued eligibility for a benefit hereunder.

In the absence of actual notice to the contrary, the Company shall make payment in accordance with the information provided by the Member. If there is a dispute as to whether a person is a Spouse, Beneficiary or other person entitled to payments hereunder, or where two or more persons make adverse claims in respect of a benefit, or where a person makes a claim that is inconsistent with information provided by the Member, the Company may obtain court directions or seek an interpleader order and the costs in respect thereof may to the extent permitted by law, be charged against the benefit to be paid.

## ARTICLE 16 GRIEVANCE PROCEDURE INFORMATION

16.01 If any dispute shall arise between the Company and any Employee, former Employee or retired Employee represented by the Union with reference to eligibility, age, Continuous Service, Credited Service or amount of retirement income or allowance, or suspension or termination of retirement income or allowance, such dispute may be taken up as a grievance under the grievance procedure of the Collective Labour Agreement then in effect between the Company and the Union omitting, however, all steps prior to the presentation of the grievance to the Human Resources Manager of the Company.

If any such grievance shall be taken to arbitration in accordance with such procedure, the appointed arbitrator or appointed board of arbitration shall have only the authority to interpret and apply the provisions of this Retirement Agreement, and of any applicable provisions of the Collective Bargaining Agreement relating to grievance procedure, but shall have no authority to alter, add to or subtract from any such provision in any way. The decision of the arbitrator or board of arbitration on any grievance properly referred shall be binding upon the Company, the Union, and such Employee, former Employee or retired Employee represented by the Union. If any dispute shall arise as to whether any such Employee, former Employee or retired Employee represented by the Union, is or continues to be permanently or totally disabled as defined in Section 5.05, such dispute shall be resolved as follows:

Such Employee, former Employee or retired Employee represented by the Union shall be examined by a physician appointed for that purpose by the Company and by a physician appointed for that purpose by the Union, and their decisions, if they shall be in agreement, shall be final and binding. If they shall disagree, the issue shall be submitted to a third physician selected by the said two physicians. The decision of such third physician after examination of the person concerned and consultation with the other two physicians appointed by the parties shall be binding upon the Company, the Union and such Employee, former Employee or retired Employee represented by the Union concerned.

The fees and expenses of any such third physician shall be borne equally by the Company and the Union.

16.02 The Union shall be furnished with such pertinent information as it may reasonably request, from time to time, concerning the operation of this Retirement Agreement and the administration of this Retirement Agreement insofar as it affects such Employee, former Employee or retired Employee represented by the Union thereunder whom it has represented.

## ARTICLE 17 UNDERTAKINGS

- 17.01 The Company undertakes that during the term of this Retirement Agreement or any extension thereof as described in Article 18, this Retirement Agreement shall, subject to Applicable Pension Laws, continue in effect without modification or change insofar as it may be applicable to Members represented by the Union, except as provided in Section 17.03.
- 17.02 The Union undertakes that during the term of this Retirement Agreement or any extension thereof, neither the Union nor any of its representatives shall:
  - (a) make any demands that this Retirement Agreement be changed in any respect or terminated or that a new Retirement Agreement be established for the Members represented by the Union or that the Company contribute or pay any greater amount for such Members than it is required to pay under the provisions of this Retirement Agreement;
  - (b) engage in or continue to engage in or in any manner encourage or sanction any strike or other action which will interfere with work or production in the Company's plant for the purpose of securing any such change, increase or termination; and
  - (c) except during the last seventy-five (75) days of the term of this Retirement Agreement or any renewal thereof the Company shall not have any obligation to negotiate or bargain with the Union with respect to any of the matters contained in the Retirement Agreement, except as provided in Section 17.03.
- **17.03** In the event that federal or provincial legislation is made effective after the Effective Date, providing for contributions

by the Company or changes in the present method of financing Company pension payments and accrual of funds, or should changes be made in the Old Age Security Act, the Company shall have the option to amend, and/or integrate the benefits provided under the Retirement Agreement by notice in writing to the Union, provided that the aggregate benefits payable to a Member under such amended agreement is not thereby reduced in amount, so that the amount such individual shall receive from the Company when added to that part of the benefits provided by any government plan which are financed or paid for by a compulsory contribution from the Company, shall not be less than the amount of benefits provided under this Retirement Agreement.

17.04 Should the Company elect not to amend, and/or integrate the benefits provided under the Retirement Agreement as provided in Section 17.03 and as a result benefits to a Member in the aggregate are lessened thereby, the Union shall have the right to re-open the Retirement Agreement for re-negotiations, with respect only to the affected benefits, sixty (60) days after the official proclamation of such change, modification, or supplementation; negotiations with respect only to the affected benefits will commence within fifteen (15) days, and if negotiations are not completed before termination of the notice period, this Retirement Agreement shall continue in full force and effect thereafter, subject to termination upon thirty (30) days written notice. Such notice shall be given only on such date as will allow for termination of this Retirement Agreement to coincide with the termination of the Collective Labour Agreement then in effect. In the event of such termination, the provisions of this Retirement Agreement shall continue to operate in the manner as provided under Applicable Pension Laws and under the Labour Relations Act of Ontario.

17.05 Notwithstanding any other provisions of the Retirement Agreement, the Company may amend the Retirement Agreement as is necessary to maintain the registration of the Retirement Agreement under Applicable Pension Laws and the Income Tax Act including an amendment providing for benefits to be reduced. Any such benefit reduction shall be subject to conditions or approval procedures under Applicable Pension Laws.

## ARTICLE 18 DURATION AND TERMINATION

- 18.01 This Retirement Agreement shall remain in full force and effect in respect to all its terms until midnight May 31, 2018, and shall continue in effect thereafter from year to year for further periods of one (1) year unless either the Company or the Union gives written notice of termination or written notice of proposals for amendment to the other party prior to, but not more than three (3) months prior to the expiry date or any yearly period thereafter. In the event of written notice having been given by either party as provided herein, negotiations shall commence within fifteen (15) days of receipt of such notice with a view to completing a new Retirement Agreement. Should such negotiations extend beyond the expiration date, this Retirement Agreement shall not expire, but shall continue in force and effect in accordance with Applicable Pension Laws and the Labour Relations Act of Ontario, or until a new Retirement Agreement is entered into by the parties, whichever date shall occur first.
- **18.02** It is understood that amendment of this Retirement Agreement is subject to the approval of the Board of Directors of the Company, subject to obtaining and retaining such acceptance of the Retirement Agreement under Applicable Pension Laws and Revenue Rules to establish that the Company is entitled to deduct the amount of its contributions to the Fund as an expense before taxes under the provisions of the Income Tax Act, or any other applicable tax laws, as now in effect or as hereafter amended or adopted. Any amendment to the terms of this Retirement Agreement shall not adversely affect any right with respect to benefits which have accrued immediately prior to the time such action is taken, except as provided in Section 18.05. The accrued benefits will be computed using as the applicable Date of Determination, the earlier of the date the Member's employment is terminated and the date of the amendment.
- 18.03 The Company agrees with the Union that, subject to the sufficiency of the assets in respect of the Retirement Agreement in the Fund and to Applicable Pension Laws and Revenue Rules, the Company will provide the retirement income provided for herein which are awarded to those who become eligible therefore during the term of this Retirement Agree-

ment or any renewal thereof. No action in performance of the terms of this Retirement Agreement and consistent herewith shall be construed or interpreted to be a violation of any of the terms of any Collective Labour Agreement between the Company and the Union.

- 18.04 Should such approval or acceptance under Applicable Pension Laws and Revenue Rules be refused, or withdrawn, negotiations will be started by the parties in an effort to make such changes as are required to obtain approval. In the event of failure in such negotiations, but in any case not later than the effective date of such refusal or withdrawal. this Retirement Agreement will be terminated. In such event, either the Union or the Company may apply to the Ontario Labour Relations Board for permission to terminate the Collective Labour Agreement then in effect (solely for the purpose of obtaining conciliation service as defined in the Collective Labour Agreement on the matter in dispute) and the other party shall join in such application. If subsequent to notice of termination of the Collective Labour Agreement or subsequent to termination of this Retirement Agreement under these circumstances, the parties settle any difference between them, then the Collective Labour Agreement, if terminated, shall be reinstated to continue in full force until termination according to its provisions.
- 18.05 In the event the Retirement Agreement shall be terminated at any time either in whole, or in part with respect to a specified group of Members only, the assets of the Fund (or the interest therein of Members affected by a partial discontinuance) shall be allocated to provide to the extent of said assets subject to Applicable Pension Laws, the retirement income and other benefits then accrued under this Retirement Agreement. The accrued benefits will be computed using the date the Member's employment is terminated as the applicable Date of Determination. Such allocation shall be made in accordance with an allocation schedule then established by the Company in consultation with the Actuary and filed with and approved by the appropriate authorities in accordance with the Applicable Pension Laws.
- 18.06 The provisions for the accrued retirement income and other benefits described in Section 18.05 may be in the form of cash or annuity contracts, or a combination thereof, at the

discretion of the Company and as permitted under Applicable Pension Laws.

18.07 Upon termination of the whole Retirement Agreement, any assets of the Fund remaining after full provision has been made for the accrued retirement income and other benefits as described in Section 18.05 may be returned to the Company or, at the option of the Company, used to increase such benefits, subject to Article 8, in such manner as the Company in its discretion shall determine, subject to Revenue Rules.

Dated at St. Jacobs, Ontario, this 29th day of May 2015
Signed on behalf of the parties hereto by their duly authorized representatives:
FOR THE COMPANY: FOR THE UNION

FOR THE COMPANY:	FOR THE UNION
R. D. Arndt Manufacturing Manager	Brias Arnold:  B. Arnold  Negotiation Committee Chair
H. Botelho Hub Manufacturing Manager	M. Perkovic Negotiation Committee Member
(	Karl Lubia F. Rubie Negotiation Committee Member
	X. Dickieson
	Negotiation Committee Member P.D. Lippert
	Negotiation Committee Member  A Bryan Staff Representative
	Starr respresentative

Printed by Thistle Printing Limited 35 Mobile Drive, Toronto, Ontario M4A 2P6

